UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM S-8

REGISTRATION STATEMENT UNDER THE SECURITIES ACT OF 1933

Main Street Capital Corporation

(Exact name of registrant as specified in its charter)

Maryland

(State or other jurisdiction of incorporation or organization)

41-2230745

(I.R.S. Employer Identification No.)

1300 Post Oak Boulevard, 8th Floor Houston, TX

77056

(Address of principal executive offices)

(Zip code)

Main Street Capital Corporation Deferred Compensation Plan

(Full title of the plan)

Vincent D. Foster Chief Executive Officer Main Street Capital Corporation 1300 Post Oak Boulevard, 8th Floor Houston, TX 77056

(Name, address and telephone number, including area code, of agent for service)

Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, or a smaller reporting company. See the definitions of "large accelerated filer," "accelerated filer" and "smaller reporting company" in Rule 12b-2 of the Exchange Act.

Large accelerated filer ⊠	Accelerated filer □	Non-accelerated filer □	Smaller reporting company □
		(Do not check if a smaller reporting company)	

CALCULATION OF REGISTRATION FEE

			Proposed		Proposed		
		Amount	Maximum		Maximum		
Title of		to be	Offering Price		Aggregate	A	mount of
Securities to be Registered	Registered Per Share Offering Price Re		Reg	Registration Fee			
Deferred compensation obligations (1)	\$	25,000,000	N/A	\$	25,000,000	\$	2,517.50
Common stock, par value \$0.01 per share		800,000(2)	29.30(3	\$)\$	23,436,000	\$	2,360.01

- (1) The deferred compensation obligations are unsecured obligations of Main Street Capital Corporation to pay deferred compensation in the future in accordance with the terms of the Main Street Capital Corporation Deferred Compensation Plan. Certain of such deferred compensation obligations may be fulfilled either in cash or in shares of Main Street Capital Corporation common stock.
- (2) These shares of common stock are to be issued under the Main Street Capital Corporation Deferred Compensation Plan. Pursuant to Rule 416(a) of the Securities Act of 1933, this Registration Statement shall be deemed to cover an indeterminate number of additional shares of common stock that become issuable under the Main Street Capital Corporation Deferred Compensation Plan by reason of any future stock dividends, stock splits or similar transactions.
- (3) Estimated solely for the purpose of calculating the registration fee pursuant to Rule 457(c) and (h) of the Securities Act, based upon the average of the high and low sales prices of Main Street Capital Corporation's common stock as reported on the New York Stock Exchange as of December 17, 2015.

TABLE OF CONTENTS

PART I PART II

Item 3. Incorporation of Documents by Reference

Item 4. Description of Securities

Item 5. Interests of Named Experts and Counsel

Item 6. Indemnification of Directors and Officers

Item 7. Exemption from Registration Claimed

Item 8. Exhibits

Item 9. Undertakings

<u>SIGNATURES</u>

EXHIBIT INDEX

Main Street Capital Corporation Deferred Compensation Plan Adoption Agreement and Plan Document

Opinion of Sutherland Asbill & Brennan LLP Consent of Grant Thornton LLP

2

Table of Contents

PART I INFORMATION REQUIRED IN THE SECTION 10(a) PROSPECTUS

The document(s) containing the information required in Part I of Form S-8 will be sent or given to participants in the Main Street Capital Corporation Deferred Compensation Plan and the Main Street Capital Corporation Deferred Compensation Plan Adoption Agreement (collectively, the "Plan") as specified by Rule 428(b)(1) under the Securities Act. In accordance with Rule 428 and the requirements of Part I of Form S-8, such documents are not being filed with the Securities and Exchange Commission (the "Commission") either as part of this registration statement or as prospectuses or prospectus supplements pursuant to Rule 424 under the Securities Act. Main Street Capital Corporation ("Main Street," "us" or "we") will maintain a file of such documents in accordance with the provisions of Rule 428. Upon request, Main Street will furnish to the Commission or its staff a copy or copies of all of the documents included in that file. These documents and the documents incorporated by reference into this registration statement on Form S-8 (this "Registration Statement") pursuant to Item 3 of Part II of this registration statement, taken together, constitute a prospectus that meets the requirements of Section 10(a) of the Securities Act.

PART II INFORMATION REQUIRED IN THE REGISTRATION STATEMENT

Item 3. Incorporation of Documents by Reference.

The following documents that have been filed with the Commission by Main Street are incorporated herein by reference and made a part hereof:

- · Main Street's Annual Report on Form 10-K for the year ended December 31, 2014;
- · Main Street's Quarterly reports on Form 10-Q for the quarters ended March 31, 2015, June 30, 2015 and September 30, 2015;
- · Main Street's Current Reports on Form 8-K filed with the Commission on March 13, 2015, April 23, 2015, April 30, 2015, May 5, 2015, May 6, 2015, June 9, 2015, August 5, 2015, October 21, 2015, November 4, 2015 and November 24, 2015 (only to the extent the information contained in each of these Forms 8-K has been filed and not furnished); and
- The description of Main Street's common stock contained in the Form 8-A filed by Main Street with the Commission on October 13, 2010, including any amendments or reports filed for the purpose of updating such description.

Each document filed with the Commission by Main Street pursuant to Section 13(a), 13(c), 14 or 15(d) of the Securities Exchange Act of 1934, as amended (the "Exchange Act"), (excluding any information furnished pursuant to Item 2.02 or Item 7.01 on any current report on Form 8-K) subsequent to the date of this registration statement and prior to the filing of a post-effective amendment to this registration statement that indicates that all securities offered hereby have been sold, or that deregisters all securities then remaining unsold, shall be deemed to be incorporated by reference in this registration statement and to be a part hereof from the date of filing of such documents.

Any statement contained in this registration statement or in a document incorporated or deemed to be incorporated herein by reference shall be deemed to be modified or superseded for purposes of this registration statement to the extent that a statement contained herein or in any subsequently filed document, which also is, or is deemed to be, incorporated by reference herein, modifies or supersedes such statement. Any such statement so modified or superseded shall not be deemed, except as so modified or superseded, to constitute a part of this registration statement.

Table of Contents

Item 4. Description of Securities.

- (a) Common Stock. Not Applicable.
- (b) Deferred Compensation Obligations.

The securities to be offered under this Registration Statement represent unsecured obligations of Main Street to pay to the participants in the Plan certain salary, bonus, board retainers and other compensation, the receipt of which the participants have elected to defer in accordance with the terms of the Plan (the "Deferred Compensation Obligations").

The Plan is a defined contribution nonqualified deferred compensation plan and is administered by Main Street. The Plan will enable certain individuals that may be designated from time to time, including non-employee directors and key employees of Main Street and its subsidiaries, to defer receipt of some or all of their cash compensation and, if applicable, restricted stock units. In addition, Main Street may also make discretionary employer contributions to the Plan. Discretionary employer contributions will be subject to a vesting schedule that will be determined by Main Street at the time of the contribution, provided that vesting will accelerate upon a change in control and the participant's death or retirement. All discretionary employer contributions, whether vested or not, will be forfeited upon a participant's termination of employment for cause or upon the participant engaging in competition with Main Street or any of its affiliates, predecessors, designees or successors.

Generally, participants will receive distributions of deferred amounts upon the earlier of separation from service, the occurrence of a disability, or a specified date (selected at the time of the deferral). Participants may elect to receive distributions in the form of a lump sum or installments, and, in some cases, may elect to delay distributions by at least five years. Participants may also be permitted to withdraw all or a portion of their deferred amounts under the Plan in the event of an unforeseeable financial emergency. Participants will receive hypothetical earnings on contributions based on returns generated by the participant's selection of investment alternatives established by Main Street, which will include a hypothetical investment in Main Street common stock. Upon distribution, participants will be paid in the form of cash, except that a participant whose account is treated as invested in shares of Main Street common stock will be paid in shares of Main Street common stock.

The foregoing does not purport to be a complete summary of the Plan and is qualified in its entirety by reference to the text of the Plan, a copy of which is included as Exhibit 4.1 to this Registration Statement and is incorporated herein by reference.

The Deferred Compensation Obligations are unsecured general obligations of Main Street and rank pari passu with other unsecured and unsubordinated indebtedness of Main Street. The Deferred Compensation Obligations may not be voluntarily or involuntarily sold, transferred, alienated, assigned, or encumbered, other than pursuant to a qualified domestic relations order.

Item 5. Interests of Named Experts and Counsel.

Not applicable.

Item 6. Indemnification of Directors and Officers.

Maryland law permits a Maryland corporation to include in its articles of incorporation a provision limiting the liability of its directors and officers to the corporation and its stockholders for money damages except for liability resulting from (a) actual receipt of an improper benefit or profit in money, property or services or (b) active and deliberate dishonesty established by a final judgment as being material to the cause of action. Our articles of incorporation contain such a provision that eliminates directors' and officers' liability to the maximum extent permitted by Maryland law, subject to the requirements of the Investment Company Act of 1940, as amended (the "1940 Act").

Our articles of incorporation require us, to the maximum extent permitted by Maryland law and subject to the requirements of the 1940 Act, to indemnify any present or former director or officer or any individual who, while a director or officer and at our request, serves or has served another corporation, real estate investment trust,

4

Table of Contents

partnership, joint venture, trust, employee benefit plan or other enterprise as a director, officer, partner or trustee, from and against any claim or liability to which such person may become subject or which such person may incur by reason of his or her service in any such capacity, except with respect to any matter as to which such person shall have been finally adjudicated in any proceeding not to have acted in good faith in the reasonable belief that his or her action was in our best interest or to be liable to us or our stockholders by reason of willful misfeasance, bad faith, gross negligence or reckless disregard of the duties involved in the conduct of such person's office.

Our bylaws obligate us, to the maximum extent permitted by Maryland law and subject to the requirements of the 1940 Act, to indemnify any present or former director or officer or any individual who, while a director or officer and at our request, serves or has served another corporation, real estate investment trust, partnership, joint venture, trust, employee benefit plan or other enterprise as a director, officer, partner or trustee and who is made, or threatened to be made, a party to a proceeding by reason of his or her service in any such capacity from and against any claim or liability to which that person may become subject or which that person may incur by reason

of his or her service in any such capacity, except with respect to any matter as to which such person shall have been finally adjudicated in any proceeding not to have acted in good faith in the reasonable belief that his or her action was in our best interest or to be liable to us or our stockholders by reason of willful misfeasance, bad faith, gross negligence or reckless disregard of the duties involved in the conduct of such person's office. Our bylaws also require that, to the maximum extent permitted by Maryland law, without requiring a preliminary determination of the ultimate entitlement to indemnification, to pay or reimburse reasonable expenses incurred by any such indemnified person in advance of the final disposition of a proceeding.

Maryland law requires a corporation (unless its articles of incorporation provide otherwise, which our articles of incorporation do not) to indemnify a director or officer who has been successful in the defense of any proceeding to which he or she is made, or threatened to be made, a party by reason of his or her service in that capacity. Maryland law permits a corporation to indemnify its present and former directors and officers, among others, against judgments, penalties, fines, settlements and reasonable expenses actually incurred by them in connection with any proceeding to which they may be made, or threatened to be made, a party by reason of his or her service in those or other capacities unless it is established that (a) the act or omission of the director or officer was material to the matter giving rise to the proceeding and (1) was committed in bad faith or (2) was the result of active and deliberate dishonesty, (b) the director or officer actually received an improper personal benefit in money, property or services or (c) in the case of any criminal proceeding, the director or officer had reasonable cause to believe that the act or omission was unlawful. However, under Maryland law, a Maryland corporation may not indemnify for an adverse judgment in a suit by or in the right of the corporation or for a judgment of liability on the basis that a personal benefit was improperly received, unless in either case a court orders indemnification, and then only for expenses. In addition, Maryland law permits a corporation to advance reasonable expenses to a director or officer upon the corporation's receipt of (a) a written affirmation by the director or officer of his or her good faith belief that he or she has met the standard of conduct necessary for indemnification by the corporation and (b) a written undertaking by him or her or on his or her behalf to repay the amount paid or reimbursed by the corporation if it is ultimately determined that the standard of conduct was not met.

In addition, we have entered into Indemnity Agreements with our directors and executive officers. The form of Indemnity Agreement entered into with each director and officer was previously filed with the Commission as Exhibit (k)(13) to our Registration Statement on Form N-2 (Reg. No. 333-142879). The Indemnity Agreements generally provide that we will, to the extent specified in the agreements and to the fullest extent permitted by the 1940 Act and Maryland law as in effect on the day the agreement is executed, indemnify and advance expenses to each indemnitee that is, or is threatened to be made, a party to or a witness in any civil, criminal or administrative proceeding. We will indemnify the indemnitee against all expenses, judgments, fines, penalties and amounts paid in settlement actually and reasonably incurred in connection with any such proceeding unless it is established that (i) the act or omission of the indemnitee was material to the matter giving rise to the proceeding and (a) was committed in bad faith or (b) was the result of active and deliberate dishonesty, (ii) the indemnitee actually received an improper personal benefit, or (iii) in the case of a criminal proceeding, the indemnitee had reasonable cause to believe his conduct was unlawful. Additionally, for so long as the we are subject to the 1940 Act, no advancement of expenses will be made until (i) the indemnitee provides a security for his undertaking, (ii) we are insured against losses arising by reason of any lawful advances, or (iii) the majority of a quorum of our disinterested directors, or independent counsel in a written opinion, determine based on a review of readily available facts that there is reason

5

Table of Contents

to believe that the indemnitee ultimately will be found entitled to indemnification. The Indemnity Agreements also provide that if the indemnification rights provided for therein are unavailable for any reason, we will pay, in the first instance, the entire amount incurred by the indemnitee in connection with any covered proceeding and waive and relinquish any right of contribution we may have against the indemnitee. The rights provided by the Indemnity Agreements are in addition to any other rights to indemnification or advancement of expenses to which the indemnitee may be entitled under applicable law, our articles of incorporation, our bylaws, any agreement, a vote of stockholders or a resolution of directors, or otherwise. No amendment or repeal of the Indemnity Agreements will limit or restrict any right of the indemnitee in respect of any action taken or omitted by the indemnitee prior to such amendment or repeal. The Indemnity Agreements will terminate upon the later of (i) ten years after the date the indemnitee has ceased to serve as our director or officer, or (ii) one year after the final termination of any proceeding for which the indemnitee is granted rights of indemnification or advancement of expenses or which is brought by the indemnitee. The above description of the Indemnity Agreements is subject to, and is qualified in its entirety by reference to, all the provisions of the form of Indemnity Agreement, previously filed with the Commission as Exhibit (k)(13) to our Registration Statement on Form N-2 (Reg. No. 333-142879).

We have obtained primary and excess insurance policies insuring our directors and officers against certain liabilities they may incur in their capacity as directors and officers. Under such policies, the insurer, on our behalf, may also pay amounts for which we have granted indemnification to the directors or officers.

Reference is made to Item 9 for our undertakings with respect to indemnification for liabilities arising under the Securities Act.

Item 7. Exemption from Registration Claimed.

Not Applicable.

Item 8. Exhibits.

The following documents are filed as a part of this registration statement or incorporated by reference herein:

Exhibit
No. Description

- *4.1 Main Street Capital Corporation Deferred Compensation Plan Adoption Agreement and Plan Document.
- *5.1 Opinion of Sutherland Asbill & Brennan LLP as to the validity of the securities being registered.
- *23.1 Consent of Grant Thornton LLP.
- *23.2 Consent of Sutherland Asbill & Brennan LLP (included in Exhibit 5.1).
- *24.1 Power of Attorney (set forth in the signature page contained in Part II of this registration statement).
- * Filed herewith.

Item 9. Undertakings.

- (a) The undersigned registrant hereby undertakes:
 - (1) To file, during any period in which offers or sales are being made, a post-effective amendment to this registration statement:
 - (i) To include any prospectus required by Section 10(a)(3) of the Securities Act of 1933;

6

Table of Contents

- (ii) To reflect in the prospectus any facts or events arising after the effective date of the registration statement (or the most recent post-effective amendment thereof) which, individually or in the aggregate, represent a fundamental change in the information set forth in the registration statement. Notwithstanding the foregoing, any increase or decrease in volume of securities offered (if the total dollar value of securities offered would not exceed that which was registered) and any deviation from the low or high end of the estimated maximum offering range may be reflected in the form of prospectus filed with the Commission pursuant to Rule 424(b) if, in the aggregate, the changes in volume and price represent no more than a 20% change in the maximum aggregate offering price set forth in the "Calculation of Registration Fee" table in the effective registration statement;
- (iii) To include any material information with respect to the plan of distribution not previously disclosed in the registration statement or any material change to such information in the registration statement;

Provided, however, that paragraphs (a)(1)(i) and (a)(1)(ii) of this section do not apply if the registration statement is on Form S-8 and the information required to be included in a post-effective amendment by those paragraphs is contained in reports filed with or furnished to the Commission by the registrant pursuant to Section 13 or Section 15(d) of the Securities Exchange Act of 1934 that are incorporated by reference in the registration statement.

- (2) That, for the purpose of determining any liability under the Securities Act of 1933, each such post-effective amendment shall be deemed to be a new registration statement relating to the securities offered therein, and the offering of such securities at that time shall be deemed to be the initial bona fide offering thereof.
- (3) To remove from registration by means of a post-effective amendment any of the securities being registered which remain unsold at the termination of the offering.
- (b) The registrant hereby undertakes that, for purposes of determining any liability under the Securities Act of 1933, each filing of the registrant's annual report pursuant to Section 13(a) or Section 15(d) of the Securities Exchange Act of 1934 (and, where applicable, each filing of an employee benefit plan's annual report pursuant to Section 15(d) of the Securities Exchange Act of 1934) that is incorporated by reference in this registration statement shall be deemed to be a new registration statement relating to the securities offered therein, and the offering of such securities at that time shall be deemed to be the initial bona fide offering thereof.
- (c) Insofar as indemnification for liabilities arising under the Securities Act of 1933 may be permitted to directors, officers and controlling persons of the registrant pursuant to the provisions described under Item 6 above, or otherwise, the registrant has been advised that in the opinion of the Securities and Exchange Commission such indemnification is against public policy as expressed in the Securities Act of 1933 and is, therefore, unenforceable. In the event that a claim for indemnification against such liabilities (other than the payment by the registrant of expenses incurred or paid by a director, officer or controlling person of the registrant in the successful defense of any action, suit or proceeding) is asserted by such director, officer or controlling person in connection with the securities being registered, the registrant will, unless in the opinion of its counsel the matter has been settled by controlling precedent, submit to a court of appropriate jurisdiction the question whether such indemnification by it is against public policy as expressed in the Securities Act of 1933 and will be governed by the final adjudication of such issue.

Pursuant to the requirements of the Securities Act of 1933, the registrant certifies that it has reasonable grounds to believe that it meets all of the requirements for filing on Form S-8 and has duly caused this registration statement to be signed on its behalf by the undersigned, thereunto duly authorized, in the City of Houston, State of Texas, on December 18, 2015.

MAIN STREET CAPITAL CORPORATION

By: /s/ Vincent D. Foster

Vincent D. Foster Chairman and Chief Executive Officer

POWER OF ATTORNEY

KNOW ALL MEN BY THESE PRESENTS, that each person whose signature appears below hereby constitutes and appoints Vincent D. Foster and Brent D. Smith, and each of them (with full power to each of them to act alone), his true and lawful attorneys-infact and agents, with full power of substitution and re-substitution, for him and on his behalf and in his name, place and stead, in any and all capacities, to sign, execute and file this registration statement under the Securities Act of 1933, as amended, and any or all amendments (including, without limitation, post-effective amendments) to this registration statement, with all exhibits and any and all documents required to be filed with respect thereto, with the Securities and Exchange Commission or any other regulatory authority, granting unto such attorneys-in-fact and agents, and each of them, full power and authority to do and perform each and every act and thing appropriate or necessary to be done in order to effectuate the same, as fully to all intents and purposes as he himself might or could do in person, hereby ratifying and confirming all that such attorneys-in-fact and agents, or any of them, or their substitute or substitutes, may lawfully do or cause to be done by virtue hereof.

Pursuant to the requirements of the Securities Act of 1933, this registration statement has been signed by the following persons in the capacities indicated on December 18, 2015.

Signature	Title
/s/ Vincent D. Foster Vincent D. Foster	Chairman and Chief Executive Officer (principal executive officer)
/s/ Brent D. Smith Brent D. Smith	Chief Financial Officer and Treasurer (principal financial officer)
/s/ Shannon D. Martin Shannon D. Martin	Vice President and Chief Accounting Officer (principal accounting officer)
/s/ Michael Appling Jr. Michael Appling Jr.	Director
/s/ Joseph E. Canon Joseph E. Canon	Director
/s/ Arthur L. French Arthur L. French	Director
/s/ J. Kevin Griffin J. Kevin Griffin	Director
/s/ John E. Jackson John E. Jackson	Director
/s/ Brian E. Lane Brian E. Lane	Director
/s/ Stephen B. Solcher Stephen B. Solcher	Director
Stephen B. Solcher	

Table of Contents

EXHIBIT INDEX

Exhibit No.		Description
*4.1	_	Main Street Capital Corporation Deferred Compensation Plan Adoption Agreement and Plan Document.
*5.1	_	Opinion of Sutherland Asbill & Brennan LLP as to the validity of the securities being registered.

- *23.1 Consent of Grant Thornton LLP.
- *23.2 Consent of Sutherland Asbill & Brennan LLP (included in Exhibit 5.1).
- *24.1 Power of Attorney (set forth in the signature page contained in Part II of this registration statement).

* Filed herewith.

NONQUALIFIED SUPPLEMENTAL

DEFERRED COMPENSATION PLAN

ADOPTION AGREEMENT

This adoption agreement and the accompanying plan document have not been approved by the Department of Labor, Internal Revenue Service, Securities Exchange Commission, or any other governmental entity. Employers may not rely on this document or the accompanying plan document to ensure any particular tax consequences with respect to the Employer's particular situation, nor do these documents constitute legal or tax advice. Pen-Cal and its employees cannot provide legal or tax advice in connection with these documents. Employers must determine the extent to which the Plan is subject to Federal or state securities laws. You should have your attorney review this document and the accompanying plan document before adopting the documents. This adoption agreement and accompanying plan document cannot be used in order to avoid penalties that may be imposed on the taxpayer.

NONQUALIFIED SUPPLEMENTAL DEFERRED COMPENSATION PLAN ADOPTION AGREEMENT

ADOPTION OF PLAN — [Select one]

- Adoption The undersigned Main Street Capital Corporation (the "Employer") hereby adopts as a Nonqualified Deferred Compensation Plan for the individuals identified in Item 5 herein the form of Plan known as the Nonqualified Supplemental Deferred Compensation Plan.
- Amendment of Previous Nonqualified Deferred Compensation Plan With "Grandfathered" Amounts (the "Employer") previously has adopted a Nonqualified Deferred Compensation Plan, known as the *Jenter name of previous planJ*, and the execution of this Adoption Agreement constitutes an amendment to that Plan, effective only for Deferrals, Contributions, earnings, gains, losses, depreciation and appreciation vested and credited thereto or debited therefrom on and after the Effective Date listed in Section 2 below, or, if otherwise determined by the Employer, on and after January 1, 2005 with respect to Plan provisions required under Section 409A of the Internal Revenue Code and the regulations thereunder. All other amounts in the plan shall be subject to the provisions of the previous plan document. This option is appropriate if the previous plan contains grandfathered amounts not subject to Section 409A of the Internal Revenue Code. Grandfathered amounts were contributed to the plan prior to January 1, 2005 under the terms of the plan in effect prior to October 4, 2004, and those plan terms have not since been materially modified. Grandfathered amounts and earnings will be administered under the terms of the prior plan document.
- Restatement of Previous Nonqualified Deferred Compensation Plan Main Street Capital Corporation (the "Employer") previously has adopted a Nonqualified Deferred Compensation Plan, known as the Main Street Capital Corporation Deferred Compensation Plan for Non-Employee Directors Jenter name of previous plan], and the execution of this Adoption Agreement constitutes a restatement of that Plan, effective as of the Effective Date listed in Section 2 below for all funds under the Plan. This option is appropriate if the previous plan does not contain "grandfathered" amounts (see description above), or if Employer wishes to apply Section 409A rules to all amounts in the plan (even pre-2005 amounts), or if previous plan has been materially modified and thus become subject to Section 409A. Notwithstanding anything in this Adoption Agreement or in the Plan to the contrary, with respect to account balances as of December 31, 2015 and any future earnings thereon, the distribution provisions of the Main Street Capital Corporation Deferred Compensation Plan for Non-Employee Directors as in effect on December 31, 2015 shall continue to apply to such amounts; provided that any such distribution provision may be further deferred in accordance with the terms of the Plan (defined below).

NAME OF PLAN

The name of this Plan as adopted by the Employer is the Main Street Capital Corporation Deferred Compensation Plan (the "Plan").

1

INDIVIDUALIZED PLAN INFORMATION

With respect to the variable features contained in the Plan, the Employer hereby makes the following selections granted under the provisions of the Plan:

1. **Adopting Entity.** The Employer adopts the Plan as:

List type of business entity (corporation, partnership, controlled group of corporations, etc.) Corporation

List each Employer adopting the Plan and Employer Identification Number (EIN):

	Name Name	of Employer: Main Street Capital Corporation of Employer: of Employer: of Employer:	EIN: EIN: EIN: EIN:	41-2230745
	Name	of Employer: h additional lists as necessary)	EIN:	
		dopting Employers and the Employer are referred to herein	collectively as the "Er	nployer."
		t state of controlling law (see Section 10.7 of Plan Docume	·	•
		State of incorporation;	,	
	X	State of domicile Texas		
2.	Effec 2016.	etive Date. The "Effective Date" of the adoption of this Pla	n, this Plan amendmen	nt or this Plan restatement is January 1,
3.	Plan	Year. The "Plan year" of the Plan shall be [select one]:		
	X	(a) the calendar year.		
		(b) the fiscal year or other 12- month period ending on	the last day of <i>[specif</i>]	w month].
		a short Plan year beginning on , and indicated in (a) or (b) above.	d ending on ,	; and thereafter the Plan year shall be as
4.	Plan	Administrator . The "Administrator" of the Plan is the Co	mpany	
		the name(s) of the individual(s) or job title(s) or entity (so Plan], and such other person(s) or entity as the Employer s		
		2		
5.	Eligil	ble Individuals. The following shall be eligible to particip	ate in the Plan: <i>[selec</i> .	t all that apply — do not list individual
	names	s]:		
	X	A select group of management or highly-compensated lor agreements;	Employees as designat	ed by the Employer in separate resolutions
		Employee Board Members;		
	X	Non-Employee Board Members;		
		Other Service Providers (i.e., independent contractors, o	consultants, etc.)	
		Employees or other Service Providers above the follow	ing Compensation thre	eshold: [enter dollar amount] \$;
		Employees with the following job titles: [enter job title	e(s); for example, "Vio	ce President and above"]
		Other: [enter description]		
6.	Based indica vs. Bo	bility Timing. Eligibility timing selected below shall apply Bonus Deferrals), as well as Employer Matching Contributed. If the Employer wishes to provide for separate eligibilionus), or for types of Contributions (for example, Employer and attach exhibits as necessary <i>[select one]</i> :	tions and Other Emplo ity rules for different t	yer Contributions, unless otherwise ypes of Compensation (for example, Salary
	X	Eligible immediately upon properly completed designate	tion by the Plan admin	istrator or Employer;
		Eligible after the following period of employment, Boa example, 90 days];	rd service, etc. <i>[enter l</i>	number of days, months or years, for
		Other [enter description]:		
7.	incren Partic	es and Amounts of Participant Deferrals [select all that a ments of one percent (for example, Salary minimum 0% m cipant's Compensation below the amount necessary to sati red Participant Contributions into another Employer-spon	aximum 100%). Not sfy required withhold	e that no Deferral election can reduce a ing for FICA/Medicare/income taxes,

repayments, etc.]:

X	Salary [select one]:
	☑ percentage <i>[enter minimum 0% and maximum 100%]</i> or
	☐ fixed dollar amount <i>[enter minimum \$]</i> .
	3
X	Non-Performance-Based Bonus [select one]:
	☑ percentage <i>[enter minimum 0% and maximum 100%]</i> or
	$\Box \text{ fixed dollar amount } \textit{[enter minimum $ $ $]}.$
X	Performance-Based Bonus <i>[select one and enter performance period (for example, 12-month period ending each March 31]</i> : performance period from <u>January 1</u> to <u>December 31</u> .
	☑ percentage [enter minimum 0% and maximum 100%]
	\square fixed dollar amount <i>[enter minimum \$]</i> .
	Commissions [select one]:
	□ percentage [enter minimum % and, maximum %]
	or □ fixed dollar amount <i>[enter minimum \$]</i> .
X	Board of Directors Fees/Retainer (note — should not include expense reimbursements):
	☑ percentage [enter minimum 0% and, maximum 100%]
	or □ fixed dollar amount <i>[enter minimum \$]</i> .
	Other Service Provider Fees or other earned income from the Employer:
	□ percentage [enter minimum % and, maximum %]
	\square fixed dollar amount <i>[enter minimum \$]</i> .
	401(k) Refund (amount deferred from Participant's regular Compensation equal in value to any refund paid to Participant in that year resulting from excess deferrals in Employer's 401(k) plan — see Subsection 2.9 of Plan document for definition.)
X	Social Security Trigger (amount deferred pursuant to an election by the Participant to defer a separate percentage of Compensation only from that portion of Compensation that exceeds the Social Security Taxable Wage Base for the upcoming year).
X	Other <i>[enter description]:</i> Dividends paid on shares of unvested Company stock. Any such deferral election must be made in the year before the year in which the unvested Company stock is granted.
X	Deferral of restricted stock units.
	4

NOTE: Special Rules for Multi-Year RSU Grants Structured To Provide For Annual Vesting of a Specified Portion of the Total Grant:

☑ Check this box if the Employer wishes to allow for deferral of restricted stock units that are structured so that a specified portion of the RSU grant vests annually (for example, an RSU grant over a four-year period vesting 25% annually). Under this type of grant, the election to defer may be made separately with respect to each portion of the grant that vests in a given year. However, each election for each portion of the grant must be made either: (i) within 30 days of the date of grant, and only if the RSU is structured so that vesting is contingent on the employee performing services for at least an additional 12 months subsequent to the election; or (ii) 12 months before the payment date of the RSU (vesting date is treated as the payment date for these purposes), but the election will not take effect for 12 months, and the subsequent payout date must be at least five years later than the previous payment date).

		Same definition of Compensation as in Employer's 401(k) or other applicable qualified retirement plan.
	X	Participant's total wages, salary, commissions, overtime, bonus, etc. for a given year which the Employer is required to report on Form W-2 or other appropriate form, (or, in the case of Board members, Board fees and retainer only, but not including expense reimbursements)(or, in the case of Other Service Providers, the Participant's total remuneration from the Employer for a given year pursuant to the agreement to provide services to the Employer), earned while the Participant is an Eligible Individual as determined by the Employer.
		Other [enter description]:
9.	Expi	ration of Participant's Deferral Elections [select all that apply]:
	X	Renewed Each Year: Participant's Deferral Elections must be renewed each year during the open enrollment period ending no later than December 31 prior to the effective Plan year (or, in the case of Performance-Based Bonuses, no less than 6 months prior to the end of the applicable performance period).
		ĭ For all types of Compensation Deferrals.
		☐ For Salary Deferrals only — other types of Deferrals are "evergreen."
		☐ For Performance-Based Bonus only — other types of Deferrals are "evergreen."
		□ Other: [specify]
		Evergreen: Participant's Deferral Elections will be "evergreen" (i.e., will continue indefinitely until the Participant's Termination Date unless changed by the
		5
		Participant — so each year the Participant will be deemed to have the same election in place as the prior year unless actively changed by the Participant during the open enrollment period ending no later than December 31 prior to the effective Plan year or, in the case of Performance-Based Bonuses, no less than 6 months prior to the end of the applicable performance period).
		☐ For all types of Compensation Deferrals.
		☐ For Salary Deferrals only — other types of Deferrals are renewed each year.
		☐ For Performance-Based Bonus only — other types of Deferrals are renewed each year.
		□ Other: [specify]
10.	Empl	oyer Contributions [select all that apply]:.
		(a) No Employer Contributions.
		(b) Matching Contributions [enter description of matching formula below and also complete Items 11 and 12]
	X	(c) Employer Contributions other than Matching Contributions [enter description of Employer Contribution formula below and complete Item 13]
		For any Plan Year, the Employer may elect to make a Discretionary Employer Contribution for any Participant
11.		oyees Eligible to Receive Employer Matching Contributions. Matching Contributions made for each Plan Year (if able) shall be allocated and credited to the Accounts of the following Participants: [Select one if applicable]
		Participants who were employed by the Employer (or, in the case of non-Employee Board Members, served on the Board) during that Plan Year, or, in the case of Other Service Providers, who provided services to the Employer during that Plan Year.
		Participants who were employed by the Employer (or, in the case of non-Employee Board Members, served on the Board) on the last day of the Plan Year, or, in the case of Other Service Providers, who provided services to the Employe on the last day of the Plan Year.
		Participants who were employed by the Employer (or, in the case of non-Employee Board Members, served on the Board) on the last day of the Plan Year or who retired, died or were Disabled during the Plan Year, or, in the case of Other Service Providers, who provided services to the Employer on the last day

definition of "Disability".] Vesting Schedule of Employer Matching Contributions. If Matching Contributions are made to the Plan, select the rate at 12. which such Contributions will vest [select one]: Immediate 100% vesting for all Participants. "Cliff" vesting (0% up to cliff; 100% after cliff) [select one]: \square 1 year cliff (less than 1 year 0%; 1 or more years 100%) □ 2 year cliff (less than 2 years 0%; 2 or more years 100%) ☐ Other cliff (*enter number of years:* less than years 0%; or more years 100%) "Graded" vesting *[enter vesting percentages]*: 1 year % 6 years 11 years % 2 years % % 12 years 7 years % 3 years % 8 years % 13 years % 4 years 9 years % 14 years 5 years % 10 years % 15 years % Other vesting schedule: [describe schedule — subject to approval] 13. Vesting Schedule of Employer Contributions (Other Than Matching Contributions). If Employer Contributions (other than Matching Contributions) are made to the Plan, select the rate at which such Contributions will vest [select one]: Immediate 100% vesting for all Participants. "Cliff" vesting (0% up to cliff; 100% after cliff) [select one]: □ 1 year cliff (less than 1 year 0%; 1 or more years 100%) ☐ 2 year cliff (less than 2 years 0%; 2 or more years 100%) ☐ Other cliff (enter number of years: less than years 0%; or more years 100%) "Graded" vesting [enter vesting percentages]: 1 year % % 6 years 11 years % 2 years 7 years 12 years % 3 years % 8 years % 13 years % % 4 years % 9 years % 14 years 5 years 10 years 15 years 7 X Other vesting schedule: [describe schedule — subject to approval] To be determined at the time of the contribution 14. Vesting Years. A "Vesting Year" described above for purposes of determining vesting under the Plan shall be computed in accordance with: [select one — if this is an amendment or restatement of a prior plan, definition from prior plan will override this definition.] Years of service (12-consecutive-month periods) with the Employer since date of hire (or date of commencement of Board service). Years of participation in the Plan (12-consecutive-month period between date Participant enters Plan and anniversary of

of the Plan Year or who died or were Disabled during the Plan Year. [If this option is selected, complete Item 29 —

		(additional fees will apply if this item is selected).
		Plan Years since each Plan Year's total Contributions were made ("rolling vesting") (additional fees will apply if this item is selected). [If this option is selected, select either (a) or (b) below:]
		☐ (a) Vesting will be credited/updated on the last day of the Plan year.
		□ (b) Vesting will be credited/updated on the anniversary of the date the Contribution is credited.
15.	Full	Vesting Upon Occurrence of Specific Event. [select all that apply]
	X	100% vesting upon Normal Retirement [describe criteria such as age (can be partial year), years of service with the Employer (must be whole years of service), or years of participation in the Plan (must be whole years of participation)]
		Retirement when Age plus Years of Service is equal to or greater than 70
		100% vesting upon Early Retirement [describe criteria such as age (must be whole years), years of service with the Employer (must be whole years of service), or years of participation in the Plan (must be whole years of participation)]
	X	100% vesting upon Death.
	\boxtimes	100% vesting upon Disability [complete Item 29 — definition of "Disability"].
	X	100% vesting upon Change in Control of the Employer [complete Items 27 and 28 — definition of "Change in Control"]
		100% vesting upon occurrence of other event: [describe event]
		8
	~ .	
16.		ice Before Plan's Establishment Excluded. Years of service earned prior to establishment of the Plan shall be disregarded urposes of determining vesting under the Plan:
		Yes (this may be elected only if this is the establishment of a new Plan).
	\boxtimes	No.
17.	vested Items	eitures for Misconduct or Violation of Non-Compete. Participants terminating employment prior to becoming 100% d will forfeit the forfeitable percentage of their Accounts as indicated in accordance with the vesting schedule selected in 12 and/or 13. Participants may also forfeit 100% of their Matching and Employer Contribution Accounts (if applicable) the following circumstances: <i>[select any that apply]</i> :
	X	Misconduct (termination for Cause). [enter definition of Misconduct or Cause below]
		Termination for "Cause" means termination by the Company of the Participant's employment with the Company or one of its subsidiaries for the Participant's (as determined by the Board in its sole discretion):
		(i) willful failure to substantially perform the Participant's duties to the Company or any of its subsidiaries;
		(ii) breach of the Participant's duty of loyalty to the Company or any of its subsidiaries;
		(iii) commission of an act of dishonesty toward the Company or any of its subsidiaries, theft of corporate property of the Company or any of its subsidiaries, or usurpation of the corporate opportunities of the Company or any of its subsidiaries;
		(iv) unethical business conduct including any violation of law connected with the Participant's employment at the Company or any of its subsidiaries; or
		(v) conviction of any felony involving dishonest or immoral conduct.
		For purposes of this definition of Cause, an act or failure to act by the Participant shall be considered "willful" only if the Participant's conduct was not in good faith and the Participant lacked a reasonable belief that the Participant's act or omission was in the best interests of the Company.
	X	Engaging in competition with the Employer. [enter definition of engaging in competition below]

"Engaging in competition" means an employee-Participant, during his or her employment with the Company or any of its subsidiaries, directly or indirectly, personally, or as an employee, officer, director, principal of or investor in, or

such date) (if this is an amendment or restatement of a prior Plan, years of participation in prior plan will be included)

or independent contractor with, another entity, engaging in business with, being employed by, or rendering any consultation or business advice or other services with respect to, any business in the United States which competes with the Company Business, as determined by the Board in its sole discretion; provided, however, that Participant may invest in (i) up to (but not more than) one percent (1%) of any class of securities of any enterprise (but without otherwise participating in the management or activities of such enterprise) if such securities are listed on any national or regional securities exchange or have been registered under Section 12(g) of the Securities Exchange Act of 1934 or (ii) up to (but not more than) ten percent (10%) of any class of securities of any private fund (but without otherwise participating in the management or activities of such enterprise). For purposes of this provision, the term "Company Business" means investing through equity capital or long-term debt in lower middle-market businesses, or investing through long-term debt in middle-market businesses.

		in middle-market businesses.
18.		oyer Stock as Deemed Investment Option. If Employer stock will be a deemed investment option, indicate below how are to be tracked: [select one]
	X	Partial and whole shares.
		Unitized fund.
		NOTE: Distributions of Employer Stock investments under the Plan shall be made in whole shares of Employer's common stock for each whole share of Employer Stock under the Plan, and in cash for any fractional share of Employer Stock under the Plan.
19.	be mad	rvice Distributions. If the Employer elects below, the Plan will allow distributions of Participant Deferral Contributions to de to Participants while they are still employed ("In-Service Distributions"), if they elect a fixed distribution date during the relection period. <i>[Select one]</i>
		No, In-Service Distributions will not be permitted.
	X	Yes, In-Service Distributions will be permitted. [select one].
		☑ For All Participant Deferral Contributions
		☐ For Participant Compensation Deferral Contributions (other than Performance-Based Bonus) only.
		☐ For Participant Performance-Based Bonus Deferral Contributions.
		☐ For Employer Contributions. [if selected, employer contributions must be 100% vested, and additional fees may apply]. If Employer wishes to limit in-service withdrawals to specific types of Employer Contributions, enter details below:
		10
		[Note — if "Yes" is elected above and the Plan will allow In-Service Distributions, please indicate if Participant will be permitted to make a "pushback" subsequent election to defer the original distribution date at least five years in accordance with Plan provisions (see subsection 9.1 of Plan document — note that election must be made 12 months prior to original distribution date and election will not take effect for 12 months) No J
	Please	e indicate the number of years a Participant must defer payment(s) until In-Service Distribution(s) may begin:
		☑ 2 Years after the Calendar Year for which the deferral is effective
		☐ Years after the Calendar Year for which the deferral is effective
	Please	indicate if separate In-Service Distribution Dates are allowed for each Type of Participant Deferral selected in Item 7:
		☐ No (single distribution date allowed per Plan Year)
		▼ Yes (requires additional tracked sources per Plan Year)
20.		rvice Distributions — Form and Timing of Payment. In-Service Distributions shall be made to Participants in the ring form: [Select one]

Lump Sums Only (allows a percentage up to 100% to be elected for distribution)

	X	Either 100% in Lump Sums or 100% in Installments.
		— if Installments are elected above, please indicate if Participant will be permitted to make a subsequent election to the installments in accordance with Plan provisions (see subsection 9.2 of Plan document) \boxtimes Yes \square No J
21.	Partic	reseeable Emergency Distributions Dates. If the Employer elects below, the Plan will allow distributions to be made to ipants while they are still employed if they meet the criteria for an unforeseeable emergency financial hardship preseeable Emergency Distributions"). Both Participant Deferral Contributions and Vested Employer Contributions can be noted in the event of an eligible Unforeseeable Emergency Distribution event. [Select one]
		No, Unforeseeable Emergency Distributions will not be permitted.
		11
	X	Yes, Unforeseeable Emergency Distributions will be permitted. <i>[select one below]</i> .
		☐ For active Participants only.
		☑ For active Participants, terminated Participants and Beneficiaries.
22.		of Distributions (at Termination of Employment or Death). Distributions will be made to Participants upon nation of Employment with the Employer or Death of the Participant as follows [select one]
		Lump sum only.
		Lump sum unless installments elected, but can only receive installments if Participant meets the following criteria [select all that apply— if item not selected below, then Participants in that category will receive lump sum only]:
		☐ Retirement [describe criteria such as age (can be partial year), years of service with the Employer (must be whole years of service), or years of participation in the Plan (must be whole years of participation)]
		☐ Early Retirement [describe criteria such as age (must be whole years), years of service with the Employer (must be whole years of service), or years of participation in the Plan (must be whole years of participation)]
		☐ Termination (other than for Misconduct, Cause or Violation of Non-Compete)
	X	Lump sum unless installments elected, and Participant may receive installments regardless of reason for Termination of Employment.
		— if Installments are elected above, please complete Item 26 and indicate if Participant will be permitted to make a quent election to change the number of installments in accordance with Plan provisions (see subsection 9.2 of Plan nent) No J
23.	Disab	ibution Upon Disability. If the Employer selects below, the Plan will allow distributions to be made to Participants upon ility but while they are still employed if they meet the criteria for Disability in Item 29 below. The form of distribution will same as for Termination of Employment, or as elected by the Participant.
		No, distribution upon Disability will not be permitted.
	X	Yes, distributions upon Disability will be permitted. [complete Item 29 — definition of "Disability"].
		12
24.	Expi	ration of Participant's Distribution Elections [select one]:
	X	Renewed Each Year: Participant's Distribution Election must be selected each year during the open enrollment period for the following year's contributions — if no new election is made, that year's contributions default to payment in the form of a lump sum. In-Service Distribution Elections must be made by participants each year.
		Evergreen: Participant's Distribution Election will be "evergreen" (i.e., will continue indefinitely for each year's contributions until the Participant's Termination Date unless changed by the Participant — so each year the Participant will be deemed to have the same distribution election in place as the prior year unless actively changed by the Participant at open enrollment, and the change will only be applicable to future contributions). In-Service Distribution Elections may not be treated as evergreen.

25. **Distributions Upon Change in Control:** If Employer elects below, distributions will be made to Participants upon Change in Control of the Employer (without a termination of employment of the Participant), as follows *[select one, and complete Items 27 and 28 below (definition of "Change in Control")]*

	[X]	No, D	istributions upon Change in Control will not be permitted.
		Yes, I	Distributions upon Change in Control will be permitted, in a lump sum only.
		Yes, I Particij	Distributions upon Change in Control will be permitted, in a lump sum or in installments as elected by the pant.
26.	Leng belov		allments (if Installment Distributions permitted in Item 20, 22 and/or Item 25 above) [indicate length
			installments over no fewer than 2 [enter minimum number of years — must be at least 2] and no more than 20 t Participant's election [enter maximum number of years].
27.			ontrol" — Dates of Distribution. Distributions upon a Change in Control shall occur upon the date that [select - see Subsection 9.9 of the Plan document for more details]:
	X		son or group acquires more than 50% of the total fair market value or voting power of the stock of the corporation definition of "corporation" in Item 28 below).
	X		son or group acquires ownership of stock of the corporation with at least 30% of the total voting power of the ation (select definition of "corporation" in Item 28 below).
			13
	X		son or group acquires assets from the corporation having a total fair market value of at least 40% of the value of all of the corporation immediately prior to such acquisition. (select definition of "corporation" in Item 28 below).
	X	appoin	ority of the corporation's board of directors is replaced during any 12-month period by directors whose timent or election is not endorsed by a majority of the board as constituted prior to the appointment or election definition of "corporation" in Item 28 below).
28.			ontrol" — Which Corporation the Change Relates. Distributions upon a Change in Control shall be made only Control relates to the corporation selected below: <i>[select all that apply]:</i>
		(a)	The corporation for whom the Participant is performing services at the time of the Change In Control event.
	X	(b)	The corporation liable for payments from the Plan to the Participant.
		(c)	A corporation that is a majority shareholder of a corporation described in (a) or (b) above.
		(d)	Any corporation in the chain of corporations in which each corporation is a majority shareholder of another corporation in the chain, ending in a corporation described in (a) or (b) above.
29.	Defi	nition of "	Disability." A Participant shall be considered "Disabled" if [select one]:
	X		on of any medically determinable physical or mental impairment which can be expected to result in death or can be to last for a continuous period of at least 12 months;
			ticipant is unable to engage in any substantial gainful activity by reason of any medically determinable physical or mpairment that can be expected to result in death or can be expected to last for a continuous period of not less than hs;
		the Par	ticipant is deemed to be totally disabled by the Social Security Administration;
		of disab	ticipant is determined to be disabled in accordance with a disability insurance program, provided that the definition ility under such disability insurance program complies with the requirements of one of the three preceding one above.
30.	to "ke empl	ey employe oyment ter	o "Key Employees" — Investment. In order to comply with Internal Revenue Code Section 409A, distributions sees" (see subsection 9.3 of the Plan Document for definition) of publicly traded companies made due to mination cannot be made within 6 months of the employment termination date. If distribution to a key employee to comply with this 6-month rule,

		Valued as of most recent Valuation Date and held at the Employer without allocation of additional gains or losses after such Valuation Date until payment can be made.
	X	Remain invested as if termination date had not occurred, then valued as of most recent Valuation Date and distributed.
31.	receipt b provides	Distributions. The Employer may elect whether distributions from a Participant's Account shall be permitted upon y the Plan Administrator of a Qualified Domestic Relations Order relating to a marital dissolution or separation that for payment of all or a portion of a Participant's Accounts to an alternate payee (spouse, former spouse, children, etc.). A below whether QDRO distributions will be permitted]:
		No, QDRO Distributions will not be permitted.
	X	Yes, QDRO Distributions will be permitted.
32.		nal Survivor Death Benefit from Life Insurance. In the event that life insurance is utilized as a funding vehicle for the Employer may wish to provide additional Survivor Benefit from the following options: [select one]
	\boxtimes	No additional Survivor Benefit offered, but rather Participant's vested Account balance.
		Face value of life insurance policy of Participant, if any.
		Greater of (a) face value of life insurance policy of Participant, if any, or (b) Participant's vested Account balance.
		Other: [enter amount or formula]
33.	Paymer	at of Plan Expenses. Plan expenses may be paid as follows: [select one]
	X	Directly by the Employer.
		Deducted from the Participant accounts and Plan's trust or other custodial account (mutual fund plans only, if applicable).
34.	threshold	nimis" Small Amount Cashouts. If selected by the Employer, Participant account balances that do not exceed a certain amount will be automatically cashed out upon the Participant's Termination of Employment or Death, as provided elect one]
	X	Yes, amounts that do not exceed a threshold dollar amount will automatically be cashed out: IRS 402(g) limit
		No, no "de minimis" small amounts will be cashed out.
		15

By signing this Adoption Agreement, the Employer certifies that it has consulted with legal counsel regarding the effects of the Plan, as applicable, on all parties. The Employer further certifies that it has and will limit participation in the Plan to a select group of management or highly compensated Employees, Board Members or Other Service Providers, as determined by the Employer in consultation with legal counsel. The Employer further certifies that it is the Employer's sole responsibility to ensure that each Participant with the right to direct deemed investments under the Plan that are based on securities issued by the Employer or a member of its controlled group (as defined in Code Section 414(b) and (c)) will receive a prospectus for any such deemed investment option based on such Employer securities.

The Employer is solely responsible for its compliance with applicable laws, including Federal and state securities and other applicable laws.

Only those elections that are completed shall be considered as provisions applicable to and forming a part of the Plan.

This Adoption Agreement may only be used in conjunction with the Plan document. All selections in the Adoption Agreement providing for customized or "other" plan provisions are subject to review for administrative feasibility, and may be subject to additional fees.

Terms used in this Adoption Agreement which are defined in the Plan document shall have the meaning given them therein.

The Employer hereby acknowledges that it is adopting this Nonqualified Supplemental Deferred Compensation Plan. Federal legislation or other changes in the law relating to nonqualified deferred compensation or other employee benefit plans may require that the Plan be amended.

* * *

The undersigned duly authorized owner, or officer of the Employer hereby executes the Plan on behalf of the Employer.

Main Street Capital Corporation

Employer

By /s/ Vincent D. Foster

Its Chief Executive Officer

16

NONQUALIFIED

SUPPLEMENTAL DEFERRED COMPENSATION PLAN

- PLAN DOCUMENT -

This document and the accompanying adoption agreement have not been approved by the Department of Labor, Internal Revenue Service, Securities Exchange Commission, or any other governmental entity. Employers may not rely on this document or the accompanying adoption agreement to ensure any particular tax consequences with respect to the Employer's particular situation, nor do these documents constitute legal or tax advice. Pen-Cal and its employees cannot provide legal or tax advice in connection with these documents. Employers must determine the extent to which the Plan is subject to Federal or state securities laws. You should have your attorney review this document and the accompanying adoption agreement before adopting the documents. This document and the accompanying adoption agreement cannot be used in order to avoid penalties that may be imposed on the taxpayer.

NONQUALIFIED SUPPLEMENTAL DEFERRED COMPENSATION PLAN - PLAN DOCUMENT -

SECTION 1 INTRODUCTION

1.1 Adoption of Plan and Purpose

This Plan is an unfunded, nonqualified deferred compensation plan. With the consent of the Employer (as defined in subsection 2.16) the plan may be adopted by executing the Adoption Agreement (as defined in subsection 2.3) in the form attached hereto. The Plan contains certain variable features which the Employer has specified in the Adoption Agreement. Only those variable features specified by the Employer in the Adoption Agreement will be applicable to the Employer.

The purpose of the Plan is to provide certain supplemental benefits under the Plan to a select group of management or highly compensated Employees of the Employer (in accordance with Sections 201, 301 and 401 of ERISA), Members of the Board(s) of the Employer, or Other Service Providers to the Employer (as defined below), and to allow such Employees, Board Members or Other Service Providers the opportunity to defer a portion of their salaries, bonuses and other compensation, subject to the terms of the Plan. Participants (and their Beneficiaries) shall have only those rights to payments as set forth in the Plan and shall be considered general, unsecured creditors of the Employer with respect to any such rights. The Plan is designed to comply with Code Section 409A and all guidance issued in connection with Code Section 409A. It is intended that the Plan be interpreted according to a good faith interpretation of Code Section 409A, and consistent with published IRS guidance, including proposed and final IRS regulations under Code Section 409A. Treatment of amounts in the Plan under any transition rules provided under all IRS and other guidance in connection with Code Section 409A shall be expressly authorized hereunder in accordance with procedures developed by the Administrator. In the event of any inconsistency between the terms of the Plan and Code Section 409A (and regulations thereunder), the terms of Code Section 409A (and the regulations thereunder) shall control. The Plan is intended to constitute an account balance plan (as defined in Treasury Regulation Section 1.409A-1(c)).

By becoming a Participant and making deferrals under this Plan, each Participant agrees to be bound by the provisions of the Plan and the determinations of the Employer and the Administrator hereunder.

1.2 Adoption of the Plan

The Employer may adopt the Plan by completing and signing the Adoption Agreement in the form attached hereto.

1.3 Plan Year

The Plan is administered on the basis of a Plan Year, as defined in subsection 2.27.

1

The plan shall be administered by a plan administrator (the "Administrator," as that term is defined in Section 3(16)(A) of ERISA) designated by the Employer in the Adoption Agreement. The Administrator has full discretionary authority to construe and interpret the provisions of the Plan and make factual determinations thereunder, including the power to determine the rights or eligibility of employees or participants and any other persons, and the amounts of their benefits under the plan, and to remedy ambiguities, inconsistencies or omissions, and such determinations shall be binding on all parties. The Administrator from time to time may adopt such rules and regulations as may be necessary or desirable for the proper and efficient administration of the Plan and as are consistent with the terms of the Plan. The administrator may delegate all or any part of its powers, rights, and duties under the Plan to such person or persons as it may deem advisable, and may engage agents to provide certain administrative services with respect to the Plan. Any notice or document relating to the Plan which is to be filed with the Administrator may be delivered, or mailed by registered or certified mail, postage pre-paid, to the Administrator, or to any designated representative of the Administrator, in care of the Employer, at its principal office.

2

SECTION 2 DEFINITIONS

2.1 Account

"Account" means all notional accounts and subaccounts maintained for a Participant in order to reflect his interest under the Plan, as described in Section 6.

2.2 Administrator

"Administrator" means the individual or individuals (if any) delegated authority by the Employer to administer the Plan, as defined in subsection 1.4.

2.3 Adoption Agreement

"Adoption Agreement" shall mean the form executed by the Employer and attached hereto, which Agreement shall constitute a part of the Plan.

2.4 Beneficiary

"Beneficiary" means the person or persons to whom a deceased Participant's benefits are payable under subsection 9.5.

2.5 Board

"Board" means the Board of Directors of the Employer (if applicable), as from time to time constituted.

2.6 Board Member

"Board Member" means a member of the Board.

2.7 Bonus

"Bonus" (also referred to herein as a "Non-Performance-Based Bonus") means an award of cash that is not a Performance-Based Bonus (as defined in subsection 2.25) that is payable to an Employee (or Board Member or Other Service Provider, as applicable) in a given year, with respect to the immediately preceding Bonus performance period, which may or may not be contingent upon the achievement of specified performance goals.

2.8 Code

"Code" means the Internal Revenue Code of 1986, as amended. Reference to a specific section of the Code shall include such section, any valid regulation promulgated thereunder, and any comparable provision of any future legislation amending, supplementing, or superseding such section.

3

2.9 Compensation

"Compensation" shall mean the amount of a Participant's remuneration from the Employer designated in the Adoption Agreement for the Plan Year (or, as determined in accordance with procedures established by the Employer, for the period during which the Participant remains an Eligible Individual). Notwithstanding the foregoing, the Compensation of an Other Service Provider (as defined in subsection 2.22) shall mean his remuneration from the Employer pursuant to an agreement to provide services to the Employer. With respect to any Participant who is a Member of the Board (if applicable), "Compensation" means all cash remuneration which, absent a deferral election under the Plan, would have otherwise been received by the Board Member in the taxable year, payable to the Board Member for service on the Board and on Board committees, including any cash payable for attendance at Board meetings and Board committee meetings, but not including any amounts constituting reimbursements of expenses to Board Members. To the extent the

Employer has designated "401(k) Refunds" in the Adoption Agreement (and to the extent elected by the Participant), an amount equal to the Participant's "401(k) Refund" shall be deferred from the Participant's Compensation otherwise payable to the Participant in the next subsequent Compensation pay period (or such later pay period in the same calendar year as the Administrator determines shall be administratively feasible), and shall be credited to the Participant's Compensation Deferral Account in accordance with subsection 4.1. For purposes of this subsection, "401(k) Refund" means any amount distributed to the applicable Participant from the Employer's qualified retirement plan intended to comply with Section 401(k) of the Code that is in excess of the maximum deferral for the prior calendar year allowable under such qualified retirement plan. Notwithstanding the foregoing, the definition of compensation for purposes of determining key employees under subsection 9.3 of the Plan shall be determined solely in accordance with subsection 9.3. To the extent not otherwise designated by the Employer in a separate document forming part of the Plan, Compensation payable after December 31 of a given year solely for services performed during the Employer's final payroll period containing December 31, is treated as Compensation payable for services performed in the subsequent year in which the non-deferred portion of the payroll payment is actually made.

2.10 Compensation Deferrals

"Compensation Deferrals" means the amounts credited to a Participant's Compensation Deferral Account pursuant to the Participant's election made in accordance with subsection 4.1.

2.11 Deferral Election

"Deferral Election" means an election by a Participant to make Compensation Deferrals or Performance-Based Bonus Deferrals in accordance with Section 4.

2.12 Disability

"Disability" for purposes of this Plan shall mean the occurrence of an event as a result of which the Participant is considered disabled, as designated by the Employer in the Adoption Agreement.

2

2.13 Effective Date

"Effective Date" means the Effective Date of the Plan, as indicated in the Adoption Agreement.

2.14 Eligible Individual

"Eligible Individual" means each Board Member, Other Service Provider, or Employee of an Employer who satisfies the eligibility requirements set forth in the Adoption Agreement, for the period during which he is determined by the Employer to satisfy such requirements.

2.15 Employee

"Employee" means a person who is employed by an Employer and is treated and/or classified by the Employer as a common law employee for purposes of wage withholding for Federal income taxes. If a person is not considered to be an Employee of the Employer in accordance with the preceding sentence, a subsequent determination by the Employer, any governmental agency, or a court that the person is a common law employee of the Employer, even if such determination is applicable to prior years, will not have a retroactive effect for purposes of eligibility to participate in the Plan.

2.16 Employer

"Employer" means the business entity designated in the Adoption Agreement, and its successors and assigns unless otherwise herein provided, or any other corporation or business organization which, with the consent of the Employer, or its successors or assigns, assumes the Employer's obligations hereunder, and any affiliate or subsidiary of the Employer or other corporation or business organization in the Employer's "controlled group" (as defined in Subsections 414(b) and (c) of the Code and Section 1.409A-1(h) of the Treasury Regulations), that has adopted the Plan on behalf of its Eligible Individuals with the consent of the Employer.

2.17 Employer Contributions

"Employer Contributions" means the amounts other than Matching Contributions that are credited to a Participant's Employer Contributions Account under the Plan by the Employer in accordance with subsection 4.4.

2.18 ERISA

"ERISA" means the Employee Retirement Income Security Act of 1974, as amended. Reference to a specific section of ERISA shall include such section, any valid regulation promulgated thereunder, and any comparable provision of any future legislation amending, supplementing, or superseding such section.

2.19 Fiscal Year Compensation

"Fiscal Year Compensation" means Compensation relating to a period of service coextensive with one or more consecutive noncalendar-year fiscal years of the Employer, where no amount of such Compensation is paid or payable during the service period. For example, a Bonus based upon a service period of two consecutive fiscal years payable after the completion of the second fiscal year would be "Fiscal Year Compensation," but periodic salary payments or Bonuses based on service periods other than the Employer's fiscal year would not be Fiscal Year Compensation.

2.20 Investment Funds

"Investment Funds" means the notional funds or other investment vehicles designated pursuant to subsection 5.1.

2.21 Matching Contributions

"Matching Contributions" means the amounts credited to a Participant's Employer Contribution Account under the Plan by the Employer that are based on the amount of Participant Deferrals made by the Participant under the Plan, or that are based upon such other formula as designated by the Employer in the Adoption Agreement, in accordance with subsection 4.3.

2.22 Other Service Providers

"Other Service Providers" shall mean independent contractors, consultants, or other similar providers of services to the Employer, other than Employees and Board Members. To the extent that an Other Service Provider is unrelated to the Employer and satisfies the other requirements of Treasury Regulation Section 1.409A-1(f)(2)(i) as described therein and in Code Section 409A and other applicable regulations, guidance, etc. thereunder, the provisions of such guidance shall not apply. To the extent that an Other Service Provider uses an accrual method of accounting for a given taxable year, amounts deferred under the Plan in such taxable year shall not be subject to Code Section 409A and other applicable guidance thereunder, notwithstanding any provision of the Plan to the contrary.

2.23 Participant

"Participant" means an Eligible Individual who meets the requirements of Section 3 and elects to make Compensation Deferrals pursuant to Section 4, or who receives Employer Contributions or Matching Contributions pursuant to subsection 4.3 or 4.4. A Participant shall cease being a Participant in accordance with subsection 3.2 herein.

6

2.24 Participant Deferrals

"Participant Deferrals" means all amounts deferred by a Participant under this Plan, including Participant Compensation Deferrals and Participant Performance-Based Bonus Deferrals.

2.25 Performance-Based Bonus

"Performance-Based Bonus" generally means Compensation where the amount of, or entitlement to, the compensation is contingent on the satisfaction of previously established organizational or individual performance criteria relating to a performance period of at least 12 consecutive months in which the Eligible Individual performs services, pursuant to rules described in Treasury Regulation Section 1.409A-1(e).

2.26 Performance-Based Bonus Deferrals

"Performance-Based Bonus Deferrals" means the amounts credited to a Participant's Compensation Deferral Account from the Participant's Performance-Based Bonus pursuant to the Participant's election made in accordance with subsection 4.2.

2.27 Plan Year

"Plan Year" means each 12-month period specified in the Adoption Agreement, on the basis of which the Plan is administered.

2.28 Retirement

"Retirement" for purposes of this Plan means the Participant's Termination Date, as defined in subsection 2.30, after attaining any age and/or service minimums with respect to Retirement or Early Retirement as designated by the Employer in the Adoption Agreement.

2.29 Spouse

"Spouse" means the person to whom a Participant is legally married under applicable state law at the earlier of the date of the Participant's death or the date payment of the Participant's benefits commenced and who is living on the date of the Participant's death.

2.30 Termination Date

"Termination Date" means (i) with respect to an Employee Participant, the Participant's separation from service (within the meaning of Section 409A of the Code and the regulations, notices and other guidance thereunder, including death or Disability) with the Employer, and any subsidiary or affiliate of the Employer as defined in Sections 414(b) and (c) of the Code and Section 1.409A-1(h) of the Treasury Regulations; (ii) with respect to a Board Member Participant, the Participant's resignation or removal from the Board (for any reason, including death or following Disability); and (iii) with respect to any Other Service Provider, the expiration of all agreements to provide services to the Employer (for any reason, including death or

7

following Disability). The date that an Employee's, Board Member's, or Other Service Provider's performance of services for all the Employers is reduced to a level less than 20% of the average level of services performed in the preceding 36-month period, shall be considered a Termination Date, and the performance of services at a level of 50% or more of the average level of services performed in the preceding 36-month period shall not be considered a Termination Date, based on the parties' reasonable expectations as of the applicable date. A Participant's Termination Date shall not be deemed to have occurred if the Employee's, Board Member's or Other Service Provider's average level of service performed in the preceding 36-month period drops below 50% but not less than 20%, unless the Employer: (i) has designated in a writing forming part of the Plan that a level between 20% and 50% will be deemed to trigger a Termination Date, and (ii) such writing was in place at or prior to the date required under Code Section 409A and the regulations and other guidance thereunder. If such designation is subsequently changed, the change must comply with the rules regarding subsequent deferrals and the acceleration of payments described in Code Section 409A and the regulations, notices, rulings and other guidance thereunder. If a Participant is both a Board Member Participant and an Employee Participant, "Termination Date" means the date the Participant satisfies both criteria (i) and (ii) above.

2.31 Valuation Date

"Valuation Date" means the last day of each Plan Year and any other date that the Employer, in its sole discretion, designates as a Valuation Date, as of which the value of an Investment Fund is adjusted for notional deferrals, contributions, distributions, gains, losses, or expenses.

2.32 Other Definitions

Other defined terms used in the Plan shall have the meanings given such terms elsewhere in the Plan.

8

SECTION 3 ELIGIBILITY AND PARTICIPATION

3.1 Eligibility

Each Eligible Individual on the Effective Date of the Plan shall be eligible to become a Participant by properly making a Deferral Election on a timely basis as described in Section 4, or, if applicable and eligible as designated by the Employer in the Adoption Agreement, by receiving a Matching Contribution or other Employer Contribution under the Plan. Each other Eligible Individual may become a Participant by making a Deferral Election on a timely basis as described in Section 4 or, if applicable and eligible as designated by the Employer in the Adoption Agreement, by receiving a Matching Contribution or other Employer Contribution under the Plan. Each Eligible Individual's decision to become a Participant by making a Deferral Election shall be entirely voluntary. The Employer may require the Participant to complete any necessary forms or other information as it deems necessary or advisable prior to permitting the Eligible Individual to commence participation in the Plan.

3.2 Cessation of Participation

If a Termination Date occurs with respect to a Participant, or if a Participant otherwise ceases to be an Eligible Individual, no further Compensation Deferrals, Performance-Based Bonus Deferrals, Matching Contributions or other Employer Contributions shall be credited to the Participant's Accounts after the Participant's Termination Date or date the Participant ceases to be eligible (or as soon as administratively feasible after the date the Participant ceases to be eligible or, if applicable, the end of the then-current Plan Year or performance period with respect to Performance-Based Bonuses), unless he is again determined to be an Eligible Individual, but the balance credited to his Accounts shall continue to be adjusted for notional investment gains and losses under the terms of the Plan and shall be distributed to him at the time and manner set forth in Section 9. An Employee, Board Member or Other Service Provider shall cease to be a Participant after his Termination Date or other loss of eligibility as soon as his entire Account balance has been distributed.

3.3 Eligibility for Matching or Employer Contributions

An Employee Participant who has satisfied the requirements necessary to become an Eligible Individual with respect to Matching Contributions as specified in the Adoption Agreement, and who has made a Compensation Deferral election pursuant to subsection 4.1 herein or who has satisfied such other criteria as specified in the Adoption Agreement, shall be eligible to receive Matching Contributions described in subsection 4.3. An Employee Participant who has satisfied the requirements necessary to become an Eligible Individual with respect to Employer Contributions other than Matching Contributions as specified in the Adoption Agreement, shall be eligible to receive Employer Contributions described in subsection 4.4.

SECTION 4 DEFERRALS AND CONTRIBUTIONS

4.1 Compensation Deferrals Other Than Performance-Based Bonus Deferrals

Each Plan Year, an Eligible Individual may elect to defer receipt of no less than the minimum and no greater than the maximum percentage or amount selected by the Employer in the Adoption Agreement with respect to each type of Compensation (other than Performance-Based Bonuses) earned with respect to pay periods beginning on and after the effective date of the election; provided, however, that Compensation earned prior to the date the Participant satisfies the eligibility requirements of Section 3 shall not be eligible for deferral under this Plan. Except as otherwise provided in this subsection, a Participant's Deferral Election for a Plan Year under this subsection must be made not later than December 31 of the preceding Plan Year (or such earlier date as determined by the Administrator) with respect to Compensation (other than Performance-Based Bonuses) earned in pay periods beginning on or after the following January 1 in accordance with rules established by the Administrator. An election to defer restricted stock units (RSUs) into the Plan must be made by one of the following deadlines: (i) the end of the calendar year prior to the date of grant of the RSU; (ii) 12 months before the payment date of the RSU (vesting date is treated as the payment date for these purposes), but the election will not take effect for 12 months, and the subsequent payout date must be at least five years later than the original payment date); (iii) within 30 days of the date of grant (but only if the RSU is structured so that vesting is contingent on the Participant performing services for at least an additional 12 months); or (iv) within 6 months of the payment (vesting) date, but only if the RSU is performance-based under Code Section 409A, and only if the performance period must be at least 12 months long and either: (a) the amount of the compensation cannot be reasonably ascertained at the time of the election, or (b) the performance requirement is still not substantially certain to be met at the time of the election. If the Employer allows for deferral of RSUs structured so that a specified portion of the RSU grant vests periodically (for example, an RSU grant over a four-year period vesting 25% annually), then the election to defer may be made separately with respect to each portion of the grant that vests in a given year, if permitted by the Employer. However, each election for each portion of the grant must be made either: (i) within thirty days of the date of grant or each anniversary thereof, and only if the RSU is structured so that vesting is contingent on the employee performing services for at least an additional 12 months subsequent to the election; or (ii) 12 months before the payment date of the RSU (vesting date is treated as the payment date for these purposes), but the election will not take effect for 12 months, and the subsequent payout date must be at least five years later than the previous payment date.

An Employee, Board Member or Other Service Provider who first becomes an Eligible Individual during a Plan Year (by virtue of a promotion, Compensation increase, commencement of employment with the Employer, commencement of Board service, execution of an agreement to provide services to an Employer, or any other reason) shall be provided enrollment documents (including Deferral Election forms) as soon as administratively feasible following such initial notification of eligibility. Such Eligible Individual must make his Deferral Elections within 30 days after first becoming an Eligible Individual, with respect to his Compensation (other than Performance-Based Bonuses) earned on or after the effective date of the Deferral Election

10

(provided, however, that if such Eligible Individual is participating in any other account balance plan maintained by the Employer or any member of the Employer's "controlled group" (as defined in subsections 414(b) and (c) of the Code), such Eligible Individual must make his Compensation Deferral Election no later than December 31 of the preceding Plan Year (or such earlier date as determined by the Administrator), or he may not elect to make Compensation Deferrals for that initial Plan Year). If an Eligible Individual does not elect to make Compensation Deferrals during that initial 30-day period, he may not later elect to make Compensation Deferrals for that year under this subsection. In the event that an Eligible Individual first becomes eligible during a Plan Year with respect to which Fiscal Year Compensation is payable, such Eligible Individual must make his Fiscal Year Compensation Deferral Election on or before the end of the fiscal year of the Employer immediately preceding the first fiscal year in which any services are performed for which the Fiscal Year Compensation is payable.

In the case of an Employee, Board Member or Other Service Provider who is rehired (or who recommences Board Service or recommences providing services to an Employer as an Other Service Provider) after having previously been an Eligible Individual, the phrase "first becomes an Eligible Individual" in the first sentence of the preceding paragraph shall be interpreted to apply only where the Eligible Individual either (i) previously received payment of his total Account balances under the Plan, or (ii) did not previously receive payment of his total Account balances under the Plan, but is rehired (or recommences Board Service or recommences providing services to an Employer as an Other Service Provider) at least 24 months after his last day as a previously Eligible Individual prior to again becoming such an Eligible Individual. In all other cases such rehired Employee, Board Member or Other Service Provider may not elect to make Compensation Deferrals until the next date determined by the Administrator with respect to Compensation earned after the following January 1. Similarly, in the case of an Employee who recommences status as an Eligible Individual for any other reason after having previously lost his status as an Eligible Individual (due to Compensation fluctuations, transfer from an ineligible location or job classification, or otherwise), the phrase "first becomes an Eligible Individual" shall be interpreted to apply only where the Eligible Individual either: (i) previously received payment of his total Account balances under the Plan, or (ii) did not previously receive payment of his total Account balances under the Plan, but regains his status as an Eligible Individual at least 24 months after his last day as a previously Eligible Individual prior to again becoming such an Eligible Individual. In all other cases such Re-Eligible Participant may not elect to make Compensation Deferrals until the next date determined by the Administrator with respect to Compensation earned after the following January 1.

An election to make Compensation Deferrals under this subsection 4.1 shall remain in effect through the last pay period commencing in the calendar year to which the election applies (except as provided in subsections 2.9 or 4.5), shall apply with respect to the applicable type of Compensation (other than Performance-Based Bonuses) to which the Deferral Election relates earned for pay

Administrator). If a Participant fails to make a Compensation Deferral election for a given Plan Year, such Participant's Compensation Deferral Election for that Plan Year shall be deemed to be zero; provided, however, that if the Employer has elected in the Adoption Agreement that a Participant's Compensation Deferral Election shall be "evergreen", then such Participant's Compensation Deferral Election shall be deemed to be identical to the most recent applicable Deferral Election on file with the Administrator with respect to the applicable type of Compensation; provided, however, that no In-Service Distribution shall be applicable to any amounts deferred in a year in which the Participant fails to make an affirmative election, and payment of such amounts for such year shall be made in accordance with his most recent election on file with the Administrator (if no election is on file, then such amounts shall be paid to him in a single lump sum).

Compensation Deferrals shall be credited to the Participant's Compensation Deferral Account as soon as administratively feasible after such amounts would have been payable to the Participant.

4.2 Performance-Based Bonus Deferrals

Each Plan Year, an Eligible Individual may elect to defer receipt of no less than the minimum and no greater than the maximum percentage or amount selected by the Employer in the Adoption Agreement with respect to Performance-Based Bonuses earned with respect to the performance period for which the Performance-Based Bonus is earned; provided, however, that the Eligible Individual performed services continuously from a date no later than the date upon which the performance criteria are established through a date no earlier than the date upon which the Eligible Individual makes a Performance-Based Bonus Deferral Election; and further provided that in no event may an election to defer Performance-Based Bonuses be made after such Bonuses have become readily ascertainable. Except as otherwise provided in this subsection, a Participant's Performance-Based Bonus Deferral Election under this subsection must be made not later than six months (or such earlier date as determined by the Administrator) prior to the end of the performance period.

An Employee, Board Member or Other Service Provider who first becomes an Eligible Individual during a Plan Year (by virtue of a promotion, Compensation increase, commencement of employment with the Employer, commencement of Board service, execution of an agreement to provide services to an Employer, or any other reason) shall be provided enrollment documents (including Deferral Election forms) as soon as administratively feasible following such initial notification of eligibility. Such Eligible Individual must make his Performance-Based Bonus Deferral Election within 30 days after first becoming an Eligible Individual (provided, however, that if such Eligible Individual is participating in any other account balance plan maintained by the Employer or any member of the Employer's "controlled group" (as defined in subsections 414(b) and (c) of the Code), such Eligible Individual must perform services continuously from a date no later than the date the performance criteria are established, and must make his Performance-Based Bonus Deferral Election no later than six months (or such earlier date as determined by the Administrator) prior to the end of the performance period, and at a time when the Performance-Based Bonus is not readily ascertainable, or he may not elect to make

12

Performance-Based Bonus Deferrals for such initial Plan Year. In the case of a Deferral Election in the first year of eligibility that is made after the beginning of the Performance-Based Bonus performance period, the Deferral Election will apply to the portion of the Performance-Based Bonus equal to the total amount of the Performance-Based Bonus for the performance period multiplied by the ratio of the number of days remaining in the performance period after the effective date of the Deferral Election over the total number of days in the Performance Period. If an Eligible Individual does not elect to make a Performance-Based Bonus Deferral during that initial 30-day period, he may not later elect to make a Performance-Based Bonus Deferral for that performance period under this subsection.

Rules relating to the timing of elections to make a Performance-Based Bonus Deferral with respect to an Employee, Board Member or Other Service Provider who becomes an Eligible Individual (due to rehire or other similar event) after having previously been an Eligible Individual shall be applied in a manner similar to rules described applicable to rehired and other Re-Eligible Participants in subsection 4.1 above.

An election to make Performance-Based Bonus Deferrals under this subsection 4.2 shall remain in effect through the end of the performance period to which the election applies (except as provided in subsection 4.5), and shall be irrevocable (provided, however, that a Participant making a Performance-Based Bonus Deferral Election under this subsection with respect to a Performance-Based Bonus that is not yet readily ascertainable, may change his election at any time prior to the first day of the six-month period ending on the last day of the performance period for which the Performance-Based Bonus Deferral Election is applicable, subject to rules established by the Administrator). If a Participant fails to make a Performance-Based Bonus Deferral Election for a given performance period, such Participant's Performance-Based Bonus Deferral Election for that performance period shall be deemed to be zero; provided, however, that if the Employer has elected in the Adoption Agreement that a Participant's Performance-Based Deferral Election shall be "evergreen", then such Participant's Performance-Based Bonus Deferral Election on file with the Administrator; provided, however, that no In-Service Distribution shall be applicable to any amounts deferred in a year in which the Participant fails to make an affirmative election, and payment of such amounts for such year shall be made in accordance with his most recent election on file with the Administrator (if no election is on file, then such amounts shall be paid to him in a single lump sum).

administratively feasible after such amounts would have been payable to the Participant.

4.3 Matching Contributions

Matching Contributions shall be determined in accordance with the formula specified in the Adoption Agreement, and shall be credited to the Employer Contribution Accounts of Participants who have satisfied the eligibility requirements for Matching Contributions specified in the Adoption Agreement. Matching Contributions under this Plan shall be credited to such Participants' Employer Contribution Accounts as soon as administratively feasible after the

13

Applicable Period selected in the Adoption Agreement, but only with respect to Participants eligible to receive such Matching Contributions as described in the Adoption Agreement.

4.4 Other Employer Contributions

Employer Contributions other than Matching Contributions shall be discretionary from year to year, and shall be credited to the Employer Contribution Accounts of Participants who have satisfied the eligibility requirements for Employer Contributions, all as determined by the Employer and documented in writing, and such writings will form part of the Plan, as specified in the Adoption Agreement. Employer Contributions under this Plan shall be credited to such Participants' Employer Contributions Accounts as soon as administratively feasible.

4.5 No Election Changes During Plan Year

A Participant shall not be permitted to change or revoke his Deferral Elections (except as otherwise described in subsections 4.1 and 4.2), except that, if a Participant's status changes such that he becomes ineligible for the Plan, the Participant's Deferrals under the Plan shall cease as described in subsection 3.2. Notwithstanding the foregoing, in the event the Employer maintains a qualified plan designed to comply with the requirements of Code Section 401(k) that requires the cessation of all deferrals in the event of a hardship withdrawal under such plan, the Participant's Deferrals under this Plan shall cease as soon as administratively feasible upon notification to the Administrator that the participant has taken such a hardship withdrawal. Notwithstanding the foregoing, if the Employer has elected in the Adoption Agreement to permit Unforeseeable Emergency Withdrawals pursuant to subsection 9.8, the Participant's Deferrals under this Plan shall cease as soon as administratively feasible upon approval by the Administrator of a Participant's properly submitted request for an Unforeseeable Emergency Withdrawal under subsection 9.8.

4.6 Crediting of Deferrals

The amount of deferrals pursuant to subsections 4.1 and 4.2 shall be credited to the Participant's Accounts as of a date determined to be administratively feasible by the Administrator.

4.7 Reduction of Deferrals or Contributions

Any Participant Deferrals or Employer Contributions to be credited to a Participant's Account under this Section may be reduced by an amount equal to the Federal or state, local or foreign income, payroll, or other taxes required to be withheld on such deferrals or contributions or to satisfy any necessary contributions under an employee welfare benefit plan described under Section 125 of the Code. A Participant shall be entitled only to the net amount of such deferral or contribution (as adjusted from time to time pursuant to the terms of the Plan). The Administrator may notify a Participant of limitations on his Deferral Election if, as a result of any election, a Participant's Compensation from the Employer would be insufficient to cover taxes, withholding, and other required deductions applicable to the Participant.

14

SECTION 5 NOTIONAL INVESTMENTS

5.1 Investment Funds

The Employer may designate, in its discretion, one or more Investment Funds for the notional investment of Participants' Accounts. The Employer, in its discretion, may from time to time establish new Investment Funds or eliminate existing Investment Funds. The Investment Funds are for recordkeeping purposes only and do not allow Participants to direct any Employer assets (including, if applicable, the assets of any trust related to the Plan). Each Participant's Accounts shall be adjusted pursuant to the Participant's notional investment elections made in accordance with this Section 5, except as otherwise determined by the Employer or Administrator in their sole discretion.

5.2 Investment Fund Elections

The Employer shall have full discretion in the direction of notional investments of Participants' Accounts under the Plan; provided, however, that if the Employer so elects in the Adoption Agreement, each Participant may elect from among the Investment Funds for the notional investment of such of his Accounts as are permitted under the Adoption Agreement from time to time in accordance with procedures established by the Employer. The Administrator, in its discretion, may adopt (and may modify from time to time) such

rules and procedures as it deems necessary or appropriate to implement the notional investment of the Participant's Accounts. Such procedures may differ among Participants or classes of Participants, as determined by the Employer or the Administrator in its discretion. The Employer or Administrator may limit, delay or restrict the notional investment of certain Participants' Accounts, or restrict allocation or reallocation into specified notional investment options, in accordance with rules established in order to comply with Employer policy and applicable law, to minimize regulated filings and disclosures, or under any other circumstances in the discretion of the Employer. Any deferred amounts subject to a Participant's investment election that must be so limited, delayed or restricted under such circumstances may be notionally invested in an Investment Fund designated by the Administrator, or may be credited with earnings at a rate determined by the Administrator, which rate may be zero. A Participant's notional investment election shall remain in effect until later changed in accordance with the rules of the Administrator. If a Participant does not make a notional investment election, all deferrals by the Participant and contributions on his behalf will be deemed to be notionally invested in the Investment Fund designated by the Employer for such purpose, or, at the Employer's election, may remain uninvested until such time as the Administrator receives proper direction, or may be credited with earnings at a rate determined by the Administrator or Employer, which rate may be zero.

5.3 Investment Fund Transfers

A Participant may elect that all or a part of his notional interest in an Investment Fund shall be transferred to one or more of the other Investment Funds. A Participant may make such notional Investment Fund transfers in accordance with rules established from time to time by the Employer or the Administrator, and in accordance with subsection 5.2.

15

SECTION 6 ACCOUNTING

6.1 Individual Accounts

Bookkeeping Accounts shall be maintained under the Plan in the name of each Participant, as applicable, along with any subaccounts under such Accounts deemed necessary or advisable from time to time, including a subaccount for each Plan Year that a Participant's Deferral Election is in effect. Each such subaccount shall reflect (i) the amount of the Participant's Deferral during that year, any Matching Contributions or Employer Contributions credited during that year, and the notional gains, losses, expenses, appreciation and depreciation attributable thereto.

Rules and procedures may be established relating to the maintenance, adjustment, and liquidation of Participants' Accounts, the crediting of deferrals and contributions and the notional gains, losses, expenses, appreciation, and depreciation attributable thereto, as are considered necessary or advisable.

6.2 Adjustment of Accounts

Pursuant to rules established by the Employer, Participants' Accounts will be adjusted on each Valuation Date, except as provided in Section 9, to reflect the notional value of the various Investment Funds as of such date, including adjustments to reflect any deferrals and contributions, notional transfers between Investment Funds, and notional gains, losses, expenses, appreciation, or depreciation with respect to such Accounts since the previous Valuation Date. The "value" of an Investment Fund at any Valuation Date may be based on the fair market value of the Investment Fund, as determined by the Administrator in its sole discretion.

6.3 Accounting Methods

The accounting methods or formulae to be used under the Plan for purposes of monitoring Participants' Accounts, including the calculation and crediting of notional gains, losses, expenses, appreciation, or depreciation, shall be determined by the Administrator in its sole discretion. The accounting methods or formulae selected by the Administrator may be revised from time to time.

6.4 Statement of Account

At such times and in such manner as determined by the Administrator, but at least annually, each Participant will be furnished with a statement reflecting the condition of his Accounts.

16

SECTION 7 VESTING

A Participant shall be fully vested at all times in his Compensation Deferral Account (if applicable). A Participant shall be vested in his Matching Contributions and/or Employer Contributions (if applicable), in accordance with the vesting schedule elected by the Employer under the Adoption Agreement. Vesting Years of Service shall be determined in accordance with the election made by the Employer in the Adoption Agreement. Amounts in a Participant's Accounts that are not vested upon the Participant's Termination Date ("forfeitures") may be used to reinstate amounts previously forfeited by other Participants who are subsequently rehired, or may be returned to the Employer, in the discretion of the Employer or the Administrator.

If a Participant has a Termination Date with the Employer as a result of the Participant's Misconduct (as defined by the Employer in the Adoption Agreement), or if the Participant engages in Competition with the Employer (as defined by the Employer in the Adoption Agreement), and the Employer has so elected in the Adoption Agreement, the Participant shall forfeit all amounts allocated to

his or her Matching Contribution Account and/or Employer Contribution Accounts (if applicable). Such forfeitures shall be returned to the Employer.

Neither the Administrator nor the Employer in any way guarantee the Participant's Account balance from loss or depreciation. Notwithstanding any provision of the Plan to the contrary, the Participant's Account balance is subject to Section 8.

Vesting Years of Service in the event of the rehire of a Participant shall be reinstated, and amounts previously forfeited by such Participants may be reinstated from forfeitures made by other Participants, or may be reinstated by the Employer.

17

SECTION 8 FUNDING

No Participant or other person shall acquire by reason of the Plan any right in or title to any assets, funds, or property of the Employer whatsoever, including, without limiting the generality of the foregoing, any specific funds, assets, or other property of the Employer. Benefits under the Plan are unfunded and unsecured. A Participant shall have only an unfunded, unsecured right to the amounts, if any, payable hereunder to that Participant. The Employer's obligations under this Plan are not secured or funded in any manner, even if the Employer elects to establish a trust with respect to the Plan. Even though benefits provided under the Plan are not funded, the Employer may establish a trust to assist in the payment of benefits. All investments under this Plan are notional and do not obligate the Employer (or its delegates) to invest the assets of the Employer or of any such trust in a similar manner.

18

SECTION 9 DISTRIBUTION OF ACCOUNTS

9.1 Distribution of Accounts

With respect to any Participant who has a Termination Date that precedes his Retirement date, an amount equal to the Participant's vested Account balances shall be distributed to the Participant (or, in the case of the Participant's death, to the Participant's Beneficiary), in the form of a single lump sum payment, or, if subsection 9.2 applies, in the form of installment payments as designated by the Employer in the Adoption Agreement. Subject to subsection 9.3 hereof, distribution of a Participant's Accounts shall be made or begin within the 90-day period following the Participant's Termination Date (provided, however, that if calculation of the amount of the payment is not administratively practicable due to events beyond the control of the Participant, the payment will be made as soon as administratively practicable for the Administrator to make such payment). Notwithstanding any provision of the Plan to the contrary, for purposes of this subsection, a Participant's Accounts shall be valued as of a Valuation Date as soon as administratively feasible preceding the date such distribution is made, in accordance with rules established by the Administrator. A Participant's Accounts may be offset by any amounts owed by the Participant to the Employer, but such offset shall not occur in excess of or prior to the date distribution of the amount would otherwise be made to the Participant.

Notwithstanding the foregoing, to the extent designated by the Employer in the Adoption Agreement, a Participant may elect, in accordance with this subsection, a distribution date for his Compensation Deferral Accounts and/or his Employer Contributions and Matching Contributions Accounts that is prior to his Termination Date (an "In-Service Distribution"). A Participant's election of an In-Service Distribution date must: (i) be made at the time of his Deferral Election for a Plan Year; and (ii) apply only to amounts deferred pursuant to that election, and any earnings, gains, losses, appreciation, and depreciation credited thereto or debited therefrom with respect to such amounts. To the extent permitted by the Employer, a Participant may elect an In-Service Distribution date with respect to Performance-Based Bonus Deferrals that is separate from an In-Service Distribution date with respect to Compensation Deferrals other than Performance-Based Bonus Deferrals for the same year, provided that the applicable In-Service Distribution date may not be earlier than the number of years designated by the Employer in the Adoption Agreement following the year in which the applicable Compensation would have been paid absent the deferral, or as further determined or limited in accordance with rules established by the Administrator. Payments made pursuant to an In-Service Distribution election shall be made in a lump sum (or, if elected by the Employer in the Adoption Agreement, any applicable other form of payment to the extent permitted by the Employer and elected by the Participant in accordance with the terms of the Plan). Each such payment shall be made as soon as administratively feasible following January 1 of the calendar year in which the payment was elected to be made, but in no event later than the end of the calendar year in which the payment was elected to be made (provided, however, that if calculation of the amount of the payment is not administratively practicable due to events beyond the control of the Participant, the payment will be made as soon as administratively practicable for the Administrator to make such payment). For purposes of such payment, the value of the

19

Participant's Accounts for the applicable Plan Year shall be determined as of a Valuation Date preceding the date that such distribution is made, in accordance with rules established by the Administrator. In the event a Participant's Termination Date occurs (or, if elected by the Employer in the Adoption Agreement, in the event a Change in Control of the Employer occurs) prior to the date the Participant had previously elected to have an In-Service Distribution payment made to him, such amount shall be paid to the Participant under the rules applicable for payment on Termination of Employment in accordance with this subsection 9.1 and subsection 9.2. Participants must make an affirmative election with respect to payment of their In-Service Distributions, and no default or evergreen election shall be

allowed with respect to In-Service Distributions.

To the extent elected by the Employer in the Adoption Agreement, Participants whose Termination Date has not yet occurred may elect to defer payment of any In-Service Distribution, provided that such election is made in accordance with procedures established by the Administrator, and further provided that any such election must be made no later than 12 calendar months prior to the previously elected In-Service Distribution Date (which for these purposes shall be January 1 of the calendar year in which the payment was elected to be made). Participants may elect any deferred payment date, but such date must be no fewer than five years from the previously elected In-Service Distribution Date (which for these purposes shall be January 1 of the calendar year in which the payment was elected to be made).

9.2 Installment Distributions

To the extent elected by the Employer in the Adoption Agreement, a Participant may elect to receive payments from his Accounts in the form of a single lump sum, as described in Section 9.1, or in annual installments over a period elected by the Employer in the Adoption Agreement. To the extent a Participant fails to make an election, the Participant shall be deemed to have elected to receive his distribution for that Plan Year in the form of a single lump sum. To the extent elected by the Employer in the Adoption Agreement, a Participant may make a separate election with respect to his Performance-Based Bonus Deferrals for each year (as adjusted for gains and losses thereon) that provides for a different method of distribution from the method of distribution he elects with respect to his Compensation Deferrals (as adjusted for gains and losses thereon) for that year. The Participant's Employer Contributions Account attributable to such year, if any (as adjusted for gains and losses thereon), shall be distributed in the same manner as his Compensation Deferral Account for such year (or in a lump sum upon his Termination Date if no election has been made).

- (a) <u>Installment Elections</u>. A Participant will be required to make his distribution election prior to the commencement of each calendar year (or, in the event of an election with respect to Performance-Based Bonuses, prior to six months before the end of the applicable performance period), or such earlier date as determined by the Administrator.
- (b) <u>Installment Payments</u>. The first installment payment shall generally be within the 90-day period following the Participant's Termination Date (provided, however, that if calculation of the amount of the payment is not administratively practicable

20

due to events beyond the control of the Participant, the payment will be made as soon as administratively practicable for the Administrator to make such payment). Succeeding payments shall generally be made by January 1 of each succeeding calendar year, but in no event later than the end of each succeeding calendar year (provided, however, that if calculation of the amount of the payment is not administratively practicable due to events beyond the control of the Participant, the payment will be made as soon as administratively practicable for the Administrator to make such payment). The amount to be distributed in each installment payment shall be determined by dividing the value of the Participant's Accounts being paid in installments as of a Valuation Date preceding the date of each distribution by the number of installment payments remaining to be made, in accordance with rules established by the Administrator. In the event of the death of the Participant prior to the full payment of his Accounts being paid in installments, payments will continue to be made to his Beneficiary in the same manner and at the same time as would have been payable to the Participant.

To the extent elected by the Employer in the Adoption Agreement, Participants who have elected payment in installments may make a subsequent election to elect payment of that amount in the form of a lump sum, if payment of installments with respect to that year's deferrals has not yet commenced. Such election must be made in accordance with procedures established by the Administrator, and any such election must be made no later than 12 calendar months prior to the originally elected payment date of the first installment. The new payment date for the installment with respect to which such election is made must be deferred to the later of: (i) five years from the date such payment would otherwise have been made, or (ii) the last payment date of the last installment with respect to that year's deferrals. To the extent elected by the Employer in the Adoption Agreement, Participants who have elected payment in installments may make a subsequent election to change the number of such installment payments so long as no acceleration of distribution payments occurs (but no fewer than the minimum number, and not to exceed the maximum number of installments elected by the Employer in the Adoption Agreement), if payment of installments with respect to that year's Deferral Elections has not yet commenced. Such election must be made in accordance with procedures established by the Administrator, and any such election must be made no later than 12 calendar months prior to the originally elected payment date of the first installment. The new payment date for any installment with respect to which such election is made must be deferred for a period of not less than five years from the date such payment would otherwise have been made. In the event payment has been elected by the Participant in the form of installments (to the extent elected by the Employer in the Adoption Agreement), each installment payment shall be considered a separately identifiable payment. In the event payment has been elected by the Participant in the form of a lump sum (or in the event payment shall be made to the Participant in the form of a lump sum under the terms of the Plan in the absence of or in lieu of the Participant's election), then the lump sum form shall be deemed to be a separately identifiable form of payment, and the Participant may make a subsequent deferral election to elect payment of that amount in the form of installments (to the extent elected by the Employer in the Adoption Agreement) in accordance with the procedures described above for changing installment payment elections. Participants will be permitted to make such a change only once with respect to any year's Deferral Elections.

Notwithstanding anything herein to the contrary, and subject to Code Section 409A, except in the case of the Participant's death, payment under the Plan shall not be made or commence as a result of the Participant's Termination Date to any Participant who is a key employee (defined below) before the date that is not less than six months after the Participant's Termination Date. For this purpose, a key employee includes a "specified employee" (as defined in Treasury Regulation Section 1.409A-1(i)) during the entire 12-month period determined by the Administrator ending with the annual date upon which key employees are identified by the Administrator, and also including any Employee identified by the Administrator in good faith with respect to any distribution as belonging to the group of identified key employees, to a maximum of 200 such key employees, regardless of whether such Employee is subsequently determined by the Employer, any governmental agency, or a court not to be a key employee. In the event amounts are payable to a key employee in installments in accordance with subsection 9.2, any installment to be paid in the first six months shall be delayed by six months, with all other installment payments payable as originally scheduled. To the extent not otherwise designated by the Employer in a separate document forming a part of the Plan applicable to all its nonqualified deferred compensation plans, the identification date for determining the Employer's key employees is each December 31 (and the new key employee list is updated and effective each subsequent April 1). To the extent not otherwise designated by the Employer in a separate document forming a part of the Plan, the definition of compensation used to determine key employee status shall be determined under Treasury Regulation Section 1.415(c)-2(a). This subsection 9.3 is applicable only with respect to Employers whose stock is publicly traded on an "established securities market" (as defined in Treasury Regulation Section 1.409A-1(k)), and is not applicable to privately held Employers unless and until such Employers become publicly traded as defined in the Treasury regulations.

9.4 Mandatory Cash-Outs of Small Amounts

If the value of a Participant's total Accounts at his Termination Date (or his death or other applicable distribution date), or at any time thereafter, together with the value of the Participant's accounts under any other account balance plan maintained by the Employer or any member of the Employer's controlled group (as defined in subsections 414(b) and (c) of the Code) is equal to or less than such amount as stated in the Adoption Agreement (which amount shall not exceed the limit described in Section 402(g) of the Code from time to time), the Accounts will be paid to the Participant (or, in the event of his death, his Beneficiary) in a single lump sum, notwithstanding any election by the Participant otherwise. Payments made under this subsection 9.4 on account of the Participant's Termination Date shall be made within the 90-day period following the Participant's Termination Date (provided, however, that if calculation of the amount of the payment is not administratively practicable due to events beyond the control of the Participant, the payment will be made as soon as administratively practicable for the Administrator to make such payment).

22

9.5 Designation of Beneficiary

Each Participant from time to time may designate any individual, trust, charity or other person or persons to whom the value of the Participant's Accounts (plus any applicable Survivor Benefit, if elected by the Employer in the Adoption Agreement) will be paid in the event the Participant dies before receiving the value of all of his Accounts. A Beneficiary designation must be made in the manner required by the Administrator for this purpose. Primary and secondary Beneficiaries are permitted. A married participant designating a Beneficiary other than his Spouse must obtain the consent of his Spouse to such designation (in accordance with rules determined by the Administrator). Payments to the Participant's Beneficiary(ies) shall be made in accordance with subsection 9.1, 9.2 or 9.4, as applicable, after the Administrator has received proper notification of the Participant's death.

A Beneficiary designation will be effective only when the Beneficiary designation is filed with the Administrator while the Participant is alive, and a subsequent Beneficiary designation will cancel all of the Participant's Beneficiary designations previously filed with the Administrator. Any designation or revocation of a Beneficiary shall be effective as only if it is received by the Administrator. Once received, such designation shall be effective as of the date the designation was executed, but without prejudice to the Administrator on account of any payment made before the change is recorded by the Administrator. If a Beneficiary dies before payment of the Participant's Accounts have been made, the Participant's Accounts shall be distributed in accordance with the Participant's Beneficiary designation and pursuant to rules established by the Administrator. If a deceased Participant failed to designate a Beneficiary, or if the designated Beneficiary predeceases the Participant, the value of the Participant's Accounts shall be payable to the Participant's Spouse or, if there is none, to the Participant's estate, or in accordance with such other equitable procedures as determined by the Administrator.

9.6 Reemployment

If a former Participant is rehired by an Employer, or any affiliate or subsidiary of the Employer described in Section 414(b) and (c) of the Code and Treasury Regulation Section 1.409A-1(h), regardless of whether he is rehired as an Eligible Individual (with respect to an Employee Participant), or a former Participant returns to service as a Board member, any payments being made to such Participant hereunder by virtue of his previous Termination Date shall continue to be made to him without regard to such rehire. If a former Participant is rehired by the Employer (with respect to an Employee Participant) or returns to service as a Board member, and in either case any payments to be made to the Participant by virtue of his previous Termination Date have not been made or commenced, any payments being made to such Participant hereunder by virtue of his previous Termination Date shall continue to be made to him without regard to such rehire or return to service. See subsections 4.1 and 4.2 of the Plan for special rules applicable to deferral elections for rehired or Re-Eligible Participants.

9.7 Special Distribution Rules

Except as otherwise provided herein and in Section 12, Account balances of Participants in this Plan shall not be distributed earlier than the applicable date or dates described in this

Section 9. Notwithstanding the foregoing, in the case of payments: (i) the deduction for which would be limited or eliminated by the application of Section 162(m) of the Code; (ii) that would violate securities or other applicable laws; or (iii) that would jeopardize the ability of the Employer to continue as a going concern in accordance with Code Section 409A and the regulations thereunder, deferral of such payments on a reasonably consistent basis for similarly situated Participants may be made by the Employer at the Employer's discretion. In the case of a payment described in (i) above, the payment must be deferred either to a date in the first year in which the Employer or Administrator reasonably anticipates that a payment of such amount would not result in a limitation of a deduction with respect to the payment of such amount under Section 162(m), or the year in which the Participant's Termination Date occurs. In the case of a payment described in (ii) or (iii) above, payment will be made at the earliest date in the first taxable year of the Employer in which the Employer or Administrator reasonably anticipates that the payment would not jeopardize the ability of the Employer to continue as a going concern in accordance with Code Section 409A and the regulations thereunder, or the payment would not result in a violation of securities or other applicable laws. Payments intended to pay employment taxes or payments made as a result of income inclusion of an amount in a Participant's Accounts as a result of a failure to satisfy Section 409A of the Code shall be permitted at the Employer or Administrator's discretion at any time and to the extent provided in Treasury Regulations under Section 409A of the Code and IRS Notice 2005-1, Q&A-15, and any applicable subsequent guidance. "Employment taxes" shall include Federal Insurance Contributions Act (FICA) tax imposed under Sections 3101, 3121(a) and 3121(v)(2) of the Code on compensation deferred under the Plan (the "FICA Amount"), the income tax imposed under Section 3401 of the Code or corresponding provisions of applicable state, local or foreign tax laws on the FICA Amount, and to pay the additional income tax under Section 3401 of the Code or corresponding provisions of applicable state, local or foreign tax laws attributable to the pyramiding Section 3401 wages and taxes. A distribution may be accelerated as may be necessary to comply with certain conflict of interest rules in accordance with Treasury Regulation Section 1.409A-3(j)(4)(iii). With respect to a subchapter S corporation, a distribution may be accelerated to avoid a nonallocation year under Code Section 409(p) in the discretion of the Employer or Administrator, provided that the amount distributed does not exceed 125 percent of the minimum amount of distribution necessary to avoid the occurrence of a nonallocation year, in accordance with Treasury Regulation Section 1.409A-3(j)(4)(x).

9.8 Distribution on Account of Unforeseeable Emergency

If elected by the Employer in the Adoption Agreement, if a Participant or Beneficiary incurs a severe financial hardship of the type described below, he may request an Unforeseeable Emergency Withdrawal, provided that the withdrawal is necessary in light of severe financial needs of the Participant. To the extent elected by the Employer in the Adoption Agreement, the ability to apply for an Unforeseeable Emergency Withdrawal may be restricted to Participants whose Termination Date has not yet occurred. Such a withdrawal shall not exceed the amount required (including anticipated taxes on the withdrawal) to meet the severe financial need and not reasonably available from other resources of the Participant (including reimbursement or compensation by insurance, cessation of deferrals under this Plan for the remainder of the Plan Year, and liquidation of the Participant's assets, to the extent liquidation itself would not cause

2.4

severe financial hardship; provided, however, that the Participant is not required to take into account for these purposes any available distribution or loan from a qualified plan or another nonqualified deferred compensation plan). Each such withdrawal election shall be made at such time and in such manner as the Administrator shall determine, and shall be effective in accordance with such rules as the Administrator shall establish and publish from time to time. Severe financial needs are limited to amounts necessary for:

- (a) A sudden unexpected illness or accident incurred by the Participant, his Spouse, Beneficiary under the Plan, or dependents (as defined in Code Section 152(a)).
- (b) Uninsured casualty loss pertaining to property owned by the Participant.
- (c) Other similar extraordinary and unforeseeable circumstances involving an uninsured loss arising from an event outside the control of the Participant.

Withdrawals of amounts under this subsection shall be paid to the Participant in a lump sum as soon as administratively feasible following receipt of the appropriate forms and information required by and acceptable to the Administrator.

9.9 Distribution Upon Change in Control

In the event of the occurrence of a Change in Control of the Employer or a member of the Employer's controlled group (as designated by the Employer in the Adoption Agreement) to the extent permitted under Section 409A of the Code and the regulations and other guidance thereunder, distributions shall be made to Participants to the extent elected by the Employer in the Adoption Agreement, in the form elected by the Participants as if a Termination Date had occurred with respect to each Participant, or as otherwise specified by the Employer in the Adoption Agreement. The Change in Control shall relate to: (i) the corporation for whom the Participant is performing services at the time of the Change in Control event; (ii) the corporation that is liable for the payment from the Plan to the Participant (or all corporations so liable if more than one corporation is liable); (iii) a corporation that is a majority shareholder of a corporation described in (i) or (ii) above; or (iv) any corporation in a chain of corporations in which each such corporation is a majority shareholder of another corporation in the chain, ending in a corporation described in (i) or (ii) above, as elected by the Employer in the Adoption Agreement. A "majority shareholder" for these purposes is a shareholder owning more than 50% of the total fair market value and total voting power of such corporation. Attribution rules described in section 318(a) of the Code apply to determine stock ownership. Stock underlying a vested option is considered owned by the individual who holds the vested option. Notwithstanding the foregoing, if a vested option is exercisable for stock that is not substantially vested (as defined in section 1.83-3(b) and (i) of the Code), the stock

25

of stock of the corporation designated in (i) through (iv) above; (vi) a person or Group acquires ownership ("effective control") of stock of the corporation with at least 30% of the total voting power of the corporation designated in (i) through (iv) above and as further limited by Treasury Regulation Section 1.409A-3(i)(5)(vi)); (vii) a majority of the board of directors of any corporation designated in (i) through (iv) above in which no other corporation is a majority shareholder is replaced during any 12-month period by directors whose appointment or election is not endorsed by a majority of the board as constituted prior to the appointment or election; or (viii) a person or Group acquires assets from the corporation designated in (i) through (iv) above having a total fair market value of at least 40% of the value of all assets of the corporation immediately prior to such acquisition; as designated by the Employer in the Adoption Agreement. For purposes of (vi) above, if any one person, or more than one person acting as a Group, is considered to own more than 50 percent of the total fair market value or total voting power of the stock of a corporation, the acquisition of additional stock by the same person or persons is not considered to cause a change in the ownership of the corporation (or to cause a change in the effective control of the corporation under (vi) above). An increase in the percentage of stock owned by any one person, or persons acting as a Group, as a result of a transaction in which the corporation acquires its stock in exchange for property will be treated as an acquisition of stock for purposes of this subsection. For purposes of (v) through (viii) above, a Change in Control shall be further limited in accordance with Treasury Regulation Sections 1.409A-3(i)(5)(v), (vi) and (vii). Distributions under this subsection shall be made as soon as administratively feasible following such Change in Control.

9.10 Supplemental Survivor Death Benefit

A supplemental survivor death benefit shall be paid to the Beneficiary of an eligible Participant who has satisfied the following criteria prior to his death:

- (a) The Participant is eligible to participate in the Plan and, at the time of his death, had a current Account balance (regardless of whether or not the Participant actually was making Compensation Deferrals at the time of his death);
- (b) The Participant was an active Employee with the Employer at the time of his death;
- (c) The Participant completed and submitted an insurance application to the Administrator; and
- (d) The Employer subsequently purchased an insurance policy on the life of the Participant, with a death benefit payable, which policy is in effect at the time of the Participant's death.

Notwithstanding any provision of this Plan or any other document to the contrary, the supplemental survivor death benefit payable pursuant to this Subsection 9.10 shall be paid only if an insurance policy has been issued on the Participant's life and such policy is in force at the time of the Participant's death and the Employer shall have no obligation with respect to the

26

payment of the supplemental survivor death benefit, or to maintain an insurance policy for any Participants.

27

SECTION 10 GENERAL PROVISIONS

10.1 Interests Not Transferable

The interests of persons entitled to benefits under the Plan are not subject to their debts or other obligations and, except as may be required by the tax withholding provisions of the Code or any state's income tax act, may not be voluntarily or involuntarily sold, transferred, alienated, assigned, or encumbered; provided, however, that a Participant's interest in the Plan may be transferable pursuant to a qualified domestic relations order, as defined in Section 414(p) of the Code to the extent designated by the Employer in the Adoption Agreement.

10.2 Employment Rights

The Plan does not constitute a contract of employment, and participation in the Plan shall not give any Employee the right to be retained in the employ of an Employer, nor any right or claim to any benefit under the Plan, unless such right or claim has specifically accrued under the terms of the Plan. The Employer expressly reserves the right to discharge any Employee at any time.

10.3 Litigation by Participants or Other Persons

If a legal action begun against the Administrator (or any member or former member thereof), an Employer, or any person or persons to whom an Employer or the Administrator has delegated all or part of its duties hereunder, by or on behalf of any person results adversely to that person, or if a legal action arises because of conflicting claims to a Participant's or other person's benefits, the cost to the Administrator (or any member or former member thereof), the Employer or any person or persons to whom the Employer or the Administrator has delegated all or part of its duties hereunder of defending the action may be charged to the extent permitted by law to the sums, if any, which were involved in the action or were payable to the Participant or other person concerned.

10.4 Indemnification

To the extent permitted by law, the Employer shall indemnify each member of the Administrator committee, and any other employee or member of the Board with duties under the Plan, against losses and expenses (including any amount paid in settlement) reasonably incurred by such person in connection with any claims against such person by reason of such person's conduct in the performance of duties under the Plan, except in relation to matters as to which such person has acted fraudulently or in bad faith in the performance of duties. Notwithstanding the foregoing, the Employer shall not indemnify any person for any expense incurred through any settlement or compromise of any action unless the Employer consents in writing to the settlement or compromise.

28

10.5 Evidence

Evidence required of anyone under the Plan may be by certificate, affidavit, document, or other information which the person acting on it considers pertinent and reliable, and signed, made, or presented by the proper party or parties.

10.6 Waiver of Notice

Any notice required under the Plan may be waived by the person entitled to such notice.

10.7 Controlling Law

Except to the extent superseded by laws of the United States, the laws of the state indicated by the Employer in the Adoption Agreement shall be controlling in all matters relating to the Plan.

10.8 Statutory References

Any reference in the Plan to a Code section or a section of ERISA, or to a section of any other Federal law, shall include any comparable section or sections of any future legislation that amends, supplements, or supersedes that section.

10.9 Severability

In case any provision of the Plan shall be held illegal or invalid for any reason, such illegality or invalidity shall not affect the remaining provisions of the Plan, and the Plan shall be construed and enforced as if such illegal and invalid provision had never been set forth in the Plan.

10.10 Action By the Employer or the Administrator

Any action required or permitted to be taken by the Employer under the Plan shall be by resolution of its Board of Directors (which term shall include any similar governing body for any Employer that is not a corporation), by resolution or other action of a duly authorized committee of its Board of Directors, or by action of a person or persons authorized by resolution of its Board of Directors or such committee. Any action required or permitted to be taken by the Administrator under the Plan shall be by resolution or other action of the Administrator or by a person or persons duly authorized by the Administrator.

10.11 Headings and Captions

The headings and captions contained in this Plan are inserted only as a matter of convenience and for reference, and in no way define, limit, enlarge, or describe the scope or intent of the Plan, nor in any way shall affect the construction of any provision of the Plan.

20

10.12 Gender and Number

Where the context permits, words in the masculine gender shall include the feminine and neuter genders, the singular shall include the plural, and the plural shall include the singular.

10.13 Examination of Documents

Copies of the Plan and any amendments thereto are on file at the office of the Employer where they may be examined by any

Participant or other person entitled to benefits under the Plan during normal business hours.

10.14 Elections

Each election or request required or permitted to be made by a Participant (or a Participant's Spouse or Beneficiary) shall be made in accordance with the rules and procedures established by the Employer or Administrator and shall be effective as determined by the Administrator. The Administrator's rules and procedures may address, among other things, the method and timing of any elections or requests required or permitted to be made by a Participant (or a Participant's Spouse or Beneficiary). All elections under the Plan shall comply with the requirements of the Uniformed Services Employment and Reemployment Rights Act of 1994, as amended ("USERRA").

10.15 Manner of Delivery

Each notice or statement provided to a Participant shall be delivered in any manner established by the Administrator and in accordance with applicable law, including, but not limited to, electronic delivery.

10.16 Facility of Payment

When a person entitled to benefits under the Plan is a minor, under legal disability, or is in any way incapacitated so as to be unable to manage his financial affairs, the Administrator may cause the benefits to be paid to such person's guardian or legal representative. If no guardian or legal representative has been appointed, or if the Administrator so determines in its sole discretion, payment may be made to any person as custodian for such individual under any applicable state law, or to the legal representative of such person for such person's benefit, or the Administrator may direct the application of such benefits for the benefit of such person. Any payment made in accordance with the preceding sentence shall be a full and complete discharge of any liability for such payment under the Plan.

10.17 Missing Persons

The Employer and the Administrator shall not be required to search for or locate a Participant, Spouse, or Beneficiary. Each Participant, Spouse, and Beneficiary must file with the Administrator, from time to time, in writing the Participant's, Spouse's, or Beneficiary's post office address and each change of post office address. Any communication, statement, or notice

30

addressed to a Participant, Spouse, or Beneficiary at the last post office address filed with the Administrator, or if no address is filed with the Administrator, then in the case of a Participant, at the Participant's last post office address as shown on the Employer's records, shall be considered a notification for purposes of the Plan and shall be binding on the Participant and the Participant's Spouse and Beneficiary for all purposes of the Plan.

If the Administrator is unable to locate the Participant, Spouse, or Beneficiary to whom a Participant's Accounts are payable, the Participant's Accounts shall be frozen as of the date on which distribution would have been completed under the terms of the Plan, and no further notional investment returns shall be credited thereto.

If a Participant whose Accounts were frozen (or his Beneficiary) files a claim for distribution of the Accounts within 7 years after the date the Accounts are frozen, and if the Administrator or Employer determines that such claim is valid, then the frozen balance that has become payable shall be paid by the Employer to the Participant or Beneficiary in a lump sum cash payment as soon as practicable thereafter. If the Administrator notifies a Participant, Spouse, or Beneficiary of the provisions of this Subsection, and the Participant, Spouse, or Beneficiary fails to claim the Participant's, Spouse's, or Beneficiary's benefits or make such person's whereabouts known to the Administrator within 7 years after the date the Accounts are frozen, the benefits of the Participant, Spouse, or Beneficiary may be disposed of, to the extent permitted by applicable law, by one or more of the following methods:

- (a) By retaining such benefits in the Plan.
- (b) By paying such benefits to a court of competent jurisdiction for judicial determination of the right thereto.
- (c) By forfeiting such benefits in accordance with procedures established by the Administrator. If a Participant, Spouse, or Beneficiary is subsequently located, such benefits may be restored (without adjustment) to the Participant, Spouse, or Beneficiary under the Plan.
- (d) By any equitable manner permitted by law under rules adopted by the Administrator.

10.18 Recovery of Benefits

In the event a Participant, Spouse, or Beneficiary receives a benefit payment from the Plan that is in excess of the benefit payment that should have been made to such Participant, Spouse, or Beneficiary, or in the event a person other than a Participant, Spouse, or Beneficiary receives an erroneous payment from the Plan, the Administrator or Employer shall have the right, on behalf of the Plan, to recover the amount of the excess or erroneous payment from the recipient. To the extent permitted under applicable law, the Administrator or Employer may, at its option, deduct the amount of such excess or erroneous payment from any future benefits payable to the applicable Participant, Spouse, or Beneficiary.

10.19 Effect on Other Benefits

Except as otherwise specifically provided under the terms of any other employee benefit plan of the Employer, a Participant's participation in this Plan shall not affect the benefits provided under such other employee benefit plan.

10.20 Tax and Legal Effects

The Employer, the Administrator, and their representatives and delegates do not in any way guarantee the tax treatment of benefits for any Participant, Spouse, or Beneficiary, and the Employer, the Administrator, and their representatives and delegates do not in any way guarantee or assume any responsibility or liability for the legal, tax, or other implications or effects of the Plan. In the event of any legal, tax, or other change that may affect the Plan, the Employer may, in its sole discretion, take any actions it deems necessary or desirable as a result of such change.

32

SECTION 11 THE ADMINISTRATOR

11.1 Information Required by Administrator

Each person entitled to benefits under the Plan must file with the Administrator from time to time in writing such person's mailing address and each change of mailing address. Any communication, statement, or notice addressed to any person at the last address filed with the Administrator will be binding upon such person for all purposes of the Plan. Each person entitled to benefits under the Plan also shall furnish the Administrator with such documents, evidence, data, or information as the Administrator considers necessary or desirable for the purposes of administering the Plan. The Employer shall furnish the Administrator with such data and information as the Administrator may deem necessary or desirable in order to administer the Plan. The records of the Employer as to an Employee's or Participant's period of employment or membership on the Board, termination of employment or membership and the reason therefor, leave of absence, reemployment, and Compensation will be conclusive on all persons unless determined to the Administrator's or Employer's satisfaction to be incorrect.

11.2 Uniform Application of Rules

The Administrator shall administer the Plan on a reasonable basis. Any rules, procedures, or regulations established by the Administrator shall be applied uniformly to all persons similarly situated.

11.3 Review of Benefit Determinations

Benefits will be paid to Participants and their beneficiaries without the necessity of formal claims. Participants or their beneficiaries, however, may make a written request to the Administrator for any Plan benefits to which they may be entitled. Participants' written request for Plan benefits will be considered a claim for Plan benefits, and will be subject to a full and fair review. If the claim is wholly or partially denied, the Administrator will furnish the claimant with a written notice of this denial. This written notice will be provided to the claimant within 90 days after the receipt of the claim by the Administrator. If notice of the denial of a claim is not furnished to the claimant in accordance with the above within 90 days, the claim will be deemed denied. The claimant will then be permitted to proceed to the review stage described in the following paragraphs.

Upon the denial of the claim for benefits, the claimant may file a claim for review, in writing, with the Administrator. The claim for review must be filed no later than 60 days after the claimant has received written notification of the denial of the claim for benefits or, if no written denial of the claim was provided, no later than 60 days after the deemed denial of the claim. The claimant may review all pertinent documents relating to the denial of the claim and submit any issues and comments, in writing, to the Administrator. If the claim is denied, the Administrator must provide the claimant with written notice of this denial within 60 days after the Administrator's receipt of the claimant's written claim for review. The Administrator's decision on the claim for review will be communicated to the claimant in writing and will include specific references to the pertinent Plan provisions on which the decision was based. If

33

the Administrator's decision on review is not furnished to the claimant within the time limitations described above, the claim will be deemed denied on review. If the claim for Plan benefits is finally denied by the Administrator (or deemed denied), then the claimant may bring suit in federal court. The claimant may not commence a suit in a court of law or equity for benefits under the Plan until the Plan's claim process and appeal rights have been exhausted and the Plan benefits requested in that appeal have been denied in whole or in part. However, the claimant may only bring a suit in court if it is filed within 90 days after the date of the final denial of the claim by the Administrator.

With respect to claims for benefits payable as a result of a Participant being determined to be disabled, the Administrator will provide the claimant with notice of the status of his claim for disability benefits under the Plan within a reasonable period of time after a complete claim has been filed, but no later than 45 days after receipt of the claim for benefits. The Administrator may request an additional 30-day extension if special circumstances warrant by notifying the claimant of the extension before the expiration of the initial

45-day period. If a decision still cannot be made within this 30-day extension period due to circumstances outside the Plan's control, the time period may be extended for an additional 30 days, in which case the claimant will be notified before the expiration of the original 30-day extension.

If the claimant has not submitted sufficient information to the Administrator to process his disability benefit claim, he will be notified of the incomplete claim and given 45 days to submit additional information. This will extend the time in which the Administrator has to respond to the claim from the date the notice of insufficient information is sent to the claimant until the date the claimant responds to the request. If the claimant does not submit the requested missing information to the Administrator within 45 days of the date of the request, the claim will be denied.

If a disability benefit claim is denied, the claimant will receive a notice which will include: (i) the specific reasons for the denial, (ii) reference to the specific Plan provisions upon which the decision is based, (iii) a description of any additional information the claimant might be required to provide with an explanation of why it is needed, and (iv) an explanation of the Plan's claims review and appeal procedures, and (v) a statement regarding the claimant's right to bring a civil action under Section 502(a) of ERISA following a denial on appeal.

The claimant may appeal a denial of a disability benefit claim by filing a written request with the Administrator within 180 days of the claimant's receipt of the initial denial notice. In connection with the appeal, the claimant may request that the Plan provide him, free of charge, copies of all documents, records and other information relevant to the claim. The claimant may also submit written comments, records, documents and other information relevant to his appeal, whether or not such documents were submitted in connection with the initial claim. The Administrator may consult with medical or vocational experts in connection with deciding the claimant's claim for benefits.

The Administrator will conduct a full and fair review of the documents and evidence submitted and will ordinarily render a decision on the disability benefit claim no later than 45

34

days after receipt of the request for review on appeal. If there are special circumstances, the decision will be made as soon as possible, but not later than 90 days after receipt of the request for review on appeal. If such an extension of time is needed, the claimant will be notified in writing prior to the end of the first 45-day period. The Administrator's final written decision will set forth: (i) the specific reasons for the decision, (ii) references to the specific Plan provisions on which the decision is based, (iii) a statement that the claimant is entitled to receive, upon request and free of charge, access to and copies of all documents, records and other information relevant to the benefit claim, and (iv) a statement regarding the claimant's right to bring a civil action under Section 502(a) of ERISA following a denial on appeal. The Administrator's decision made in good faith will be final and binding.

11.4 Administrator's Decision Final

Benefits under the Plan will be paid only if the Administrator decides in its sole discretion that a Participant or Beneficiary (or other claimant) is entitled to them. Subject to applicable law, any interpretation of the provisions of the Plan and any decisions on any matter within the discretion of the Administrator made by the Administrator or its delegate in good faith shall be binding on all persons. A misstatement or other mistake of fact shall be corrected when it becomes known and the Administrator shall make such adjustment on account thereof as it considers equitable and practicable.

35

SECTION 12 AMENDMENT AND TERMINATION

While the Employer expects and intends to continue the Plan, the Employer and the Administrator each reserve the right to amend the Plan at any time and for any reason, including the right to amend this Section 12 and the Plan termination rules herein; provided, however, that each Participant will be entitled to the amount credited to his Accounts immediately prior to such amendment. The power to amend the Plan includes (without limitation) the power to change the Plan provisions regarding eligibility, contributions, notional investments, vesting, and distribution forms, and timing of payments, including changes applicable to benefits accrued prior to the effective date of any such amendment; provided, however, that amendments to the Plan (other than amendments relating to Plan termination) shall not cause the Plan to provide for acceleration of distributions in violation of Section 409A of the Code and applicable regulations thereunder.

The Employer reserves the right to terminate the Plan at any time and for any reason; provided, however, that each Participant will be entitled to the amount credited to his Accounts immediately prior to such termination (as adjusted for notional income, losses, expenses, appreciation and depreciation occurring from the date of such termination until the date of distribution).

In the event that the Plan is terminated pursuant to this Section 12, the balances in affected Participants' Accounts shall be distributed at the time and in the manner set forth in Section 9. Notwithstanding the foregoing, the Employer and the Administrator reserve the right to make all such distributions within the second twelve-month period commencing with the date of termination of the Plan; provided, however, that no such distribution will be made during the first twelve-month period following such date of Plan termination other than those that would otherwise be payable under Section 9 absent the termination of the Plan. In the event of a Plan termination due to a Change in Control of the Employer, distributions shall be made within 12 months of the date of the termination of the Plan.

[Letterhead of Sutherland Asbill & Brennan LLP]

December 18, 2015

Main Street Capital Corporation 1300 Post Oak Boulevard, 8th Floor Houston, TX 77056

> Re: Main Street Capital Corporation Registration Statement on Form S-8

Ladies and Gentlemen:

We have acted as counsel to Main Street Capital Corporation, a Maryland corporation (the "Company"), in connection with the preparation and filing by the Company with the Securities and Exchange Commission of a registration statement on Form S-8 (the "Registration Statement") under the Securities Act of 1933, as amended (the "Securities Act"), with respect to the offer and sale of up to (i) \$25,000,0000 of deferred compensation obligations (the "Deferred Compensation Obligations") pursuant to the Main Street Capital Corporation Deferred Compensation Plan (the "Plan") and (ii) 800,000 shares of the Company's common stock, par value \$0.01 per share ("Common Stock"), pursuant to the Plan.

As counsel to the Company, we have participated in the preparation of the Registration Statement and have examined originals or copies, certified or otherwise identified to our satisfaction by public officials or officers of the Company as authentic copies of originals, of

- (i) The Company's Articles of Amendment and Restatement, certified as of the date hereof by an officer of the Company,
- (ii) The Company's Amended and Restated Bylaws, certified as of the date hereof by an officer of the Company,
- (iii) A certificate from the Maryland State Department of Assessments and Taxation as to the good standing of the Company, as of a recent date;
- (iv) Resolutions of the board of directors of the Company relating to (a) the authorization and approval of the preparation and filing of the Registration Statement, (b) the authorization, issuance, offer and sale of the Deferred Compensation Obligations pursuant to the Registration Statement and the Plan and (c) the authorization, issuance, offer and sale of the Common Stock pursuant to the Registration Statement and the Plan, and
- (v) Such other documents or matters of law as in our judgment were necessary to enable us to render the opinions expressed below.

With respect to such examination and our opinion expressed herein, we have assumed, without any independent investigation or verification (i) the genuineness of all signatures on all documents submitted to us for examination, (ii) the legal capacity of all natural persons, (iii) the authenticity of all documents submitted to us as originals, (iv) the conformity to original documents of all documents submitted to us as conformed or reproduced copies and the authenticity of the originals of such copied documents, and (v) that all certificates issued by public officials have been properly issued. We also have assumed (i) without independent investigation or verification the accuracy and completeness of all corporate records made available to us by the Company; (ii) the Deferred Compensation Obligations are evidenced by a writing that, to the extent applicable, constitutes the legal, valid and binding obligation of each party thereto other than the Company, enforceable against such party in accordance with its terms and (iii) the Trust Indenture Act of 1939, as amended, does not apply to the Deferred Compensation Obligations.

Where factual matters material to this opinion letter were not independently established, we have relied upon certificates and/or representations of officers of the Company. We have also relied on certificates of public officials. Except as otherwise stated herein, we have not independently established the facts, or in the case of certificates of public officials, the other statements, so relied upon.

This opinion is limited to the General Corporation Law of the State of Maryland and the laws of the State of Texas, as in effect on the date hereof, and we express no opinion with respect to any other laws of the State of Maryland or the laws of any other jurisdiction. We express no opinion as to any state securities or broker-dealer laws or regulations thereunder relating to the offer, issuance and sale of the Deferred Compensation Obligations or Common Stock pursuant to the Registration Statement and the Plan.

Based on the foregoing, and subject to the qualifications stated herein, we are of the opinion that, assuming that the Plan is governed by the laws of the State of Texas, (i) the Deferred Compensation Obligations, upon being issued under and in conformity with the Plan, will be legally binding obligations of the Company enforceable in accordance with the terms of the Plan, (A) except as enforcement thereof may be limited in bankruptcy, insolvency, reorganization, moratorium or other laws of general application relating to or affecting enforcement of creditors' rights or by general principles of equity (regardless of whether such enforceability is considered a proceeding at law or in equity) and (B) assuming that, at the time of each issuance of the Deferred Compensation Obligations and after giving effect thereto, the Company will be in compliance with Section 18(a) of the Investment Company Act of 1940, as amended (the "1940 Act"), giving effect to Section 61(a)(1) of the 1940 Act and (ii) the shares of Common Stock issuable pursuant to the Registration Statement and the Plan will be, when issued and paid for in accordance with the Plan, validly issued, fully paid and nonassessable.

This opinion is limited to the matters expressly set forth herein, and no opinion may be implied or inferred beyond those expressly stated. Our opinions and other statements expressed herein are as of the date hereof, and we have no obligation to update this opinion letter or to advise you of any changes in applicable law or any other matters that may come to our attention after the date hereof.

We hereby consent to the filing of this opinion letter as an exhibit to the Registration Statement. We do not admit by giving this consent that we are in the category of persons whose consent is required under Section 7 of the Securities Act.

Respectfully submitted,

/s/ Sutherland Asbill & Brennan LLP

CONSENT OF INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM

We have issued our reports dated February 27, 2015 with respect to the consolidated financial statements, financial highlights, and Schedule 12-14 of Main Street Capital Corporation included in the Annual Report on Form 10-K for the year ended December 31, 2014, which are incorporated by reference in this Registration Statement. We consent to the incorporation by reference of the aforementioned reports in this Registration Statement.

/s/ GRANT THORNTON LLP

Dallas, Texas December 18, 2015